

# FUTURE ALTERNATIVES

Joel C. Snell Dept. of Sociology Dana College

R. Gary Dean Dept. of Business -Accounting Creighton University In FUTURE SHOCK, Alvin Toffler uses terms that now seem very different, but at the time of this writing 1973-1975 were synonymous. They are Toffler's SUPER INDUSTRIALISM and the common usage today of POST INDUSTRIALISM.

At the time, Toffler meant that SUPER INDUSTRIALISM was the third wave of society. Corporations that made goods like cars would be sent overseas. The third wave society is based on knowledge and knowledge workers. One needed numerous skills and intelligence to survive. Today, that is generally POST INDUSTRIALISM.

The theory has held up but in a negative way for the United States. Two out of three jobs that are created in this country are service workers who make something near or around minimum wage. The other job is one that is well paid, but requires a great deal of education. That means usually concrete oriented thinking, math oriented, and technological in emphasis. At the time this is written, in terms of job creation, the job most likely to emerge is a clerk and the salary of \$20,000 a year. Job advancement is limited.

We have used the two terms intermittently as was the practice of the early to mid 70's.

Further, FUTURE SHOCK is like CULTURE SHOCK. One is overwhelmed with the changes confronted and disoriented and depressed about the future. In CULTURE SHOCK, one is overwhelmed by a new culture and feels disoriented and depressed.

Our views are similar to Toffler's, and one can read our summary. Please take into account that this was finished 35 years ago. We are accountable for both what really happened and indicated in the book and our mistakes.

- JOEL C. SNELL / Professor Emeritus Kirkwood College
- R. GARY DEAN/ former Professor of Creighton University.
- SEPTEMBER-OCTOBER/ 2010

#### ABSTRACT

The focus of the study deals with the socio-economic impact on the life styles to the year 2020. The importance of life styles is viewed in terms of how living patterns retard or encourage urban spread.

Chapter One deals with a discussion of the field of futurism and some assumptions made by the researchers.

Chapter Two deals with sociological implications of man to land relations.

Chapter Three deals with urban systems encompassing the sociological and human biological aspects of urban growth.

Chapter Four deals with political aspects of population spread and Chapter Five with the economic implications.

Chapter Six deals with specific, but highly visible considerations of housing, and the Seventh Chapter is concerned with transportation.

Chapter Eight summarizes the major elements of the study.

The findings include the following. The authors used three alternative futures. One future, and the basic one discussed, "Super Industrialism," predicts a future based on projections of current trends. Basically, this scenario envisions a highly technological society and a population which is moderately increased and widely distributed about a city. The "Post Industrial Society" is one in which

there is a large consumption of energy, dominance of the individual auto, and the preponderance of individual family housing patterns. The majority of new growth in the city is at its fringes.

A second future, "The Green Revolution," (or a planned community), envisions a system dramatically different than current trends. It is based upon a post-Keynesian government in which urban growth is controlled and central cities are revitalized with modern mass transit systems, new multidwelling housing and green belts. It also assumes a reduction of tensions among social classes and ethnic groups. Population increases would be reduced to zero level growth or less. Domestic expenditures at the federal level would take priority over defense and armament expenditures. Other characteristics include the preservation of small farm communities, introduction of new towns, and the promotion of satellite cities.

A third alternative was entitled "Restoration City,"

Under this concept a combination of promotion of the market system and the enhancement of nation and local public security systems prevail as there are major fluctuations in the market. Due to severe economic downturns, households double up as grandparents, parents, and children live under the same roof. Some type of mass transit system is necessary as increasingly large portions of the population are not able to obtain credit for suburban housing and auto purchases.

Traditional social, religious, and political values are implemented; large populations may occur. These added numbers may be used in expansion-oriented national policies as federal officials select to destabilize neighboring and foreign territories to encourage outmigration to these newly annexed areas and reduce population pressure. In terms of city growth, suburbs would flourish but outward pressure would be reduced by population impaction of numerous middle and lower income residents.

The authors did not make or assert preference about the future alternatives. Rather, they suggest that the "super industrial state" is the most probable.

#### Preface

Generations write each new chapter to the future.

Here we have attempted to compile and synthesize forecasts of coming events.

Each chapter selects a discipline and searches the future in that discipline.

The first chapter describes the field of futurology and the remaining chapters review the future from the perspectives of sociology, demography, political science, and economics. Two special chapters are also included dealing with housing and transportation.

There is a brief summation in which certain conclusions are suggested, but any major conclusions are left to the reader.

The presentation is directed toward socio-political considerations. One of the major questions implicitly raised is whether or not the country will move forward with centralist tendencies or rather drift dramatically

to the right or left. Recognizing this, we have directed some attention to a U.S.A. that might experience major depression or military rule. Omitted however is reference to total, or near total, nuclear destruction. For, given this scenario, little planning or forecasting would be required.

We have also attempted to present, via footnotes and bibliography, as complete a reference source in these areas as probably exists today. We wish to thank Mrs. Beard of World Futures Society and Dr. John McHale of the Center for Integrative Studies at the University of New York (Binghamton) for their special assistance, and our typist Ms. Tricia Thennes. The authors, however, accept the responsibility for any errors, whether in judgment, omission, or suggestion.

### Table of Contents

#### Preface

- Chapter 1 Futurology: A State of the Field
- Chapter 2 Social Implications
- Chapter 3 Population and Human Ecology
- Chapter 4 Geo-Politics
- Chapter 5 The Economic Outlook
- Chapter 6 Sociology of Housing
- Chapter 7 Transportation Systems
- Chapter 8 A Summation

Chapter One

Futurology: A State

of the Field

Futurology: A State of the Field

Daniel Bell, an American futurist, notes that: "the year 2000 has already arrived, for in the decisions we make now, in the way we design our environment, and thus sketch our lines of constraint, the future is committed"1. Though social and technical forecasting are filled with jargon that connotes contemporary quantification, empiricism, and future model building, the field is an ancient art. Bell's comment not only notes the importance of the future of forecasting but also indicates the validity of futurism as it has been used in the past. In social institutions the old can structure the new. Each generation makes commitments consciously or unconsciously that give rise to new behaviors and new physical structures. Early data on preliterate systems indicates that magic, science, and religion all competed in differing ways to project the future.2

Throughout history, prediction has come from all sectors of the population, from visionaries to entrenched

establishmentarians. On short range projects, prophets, especially those who have committed their forecast to print, have had to enjoy or suffer the consequences of their prediction.

Gerald Clarke<sup>3</sup> has noted that the field of prophecy has generally not been fruitful. Most predictions have been proven wrong and the occupation of forecasting has been risky. The field is full of faulty forecasts. The Library of Congress (April 10, 1967) has gathered a large collection of poor judgments in technical, economic, and social forecasting. Others have come to similar conclusions.<sup>4</sup>

In terms of the social sciences, one of the first forecasters was an essayist and philosopher, Thomas Malthus. Malthus' prediction was of plight and misery for the lower classes in England. He based his projections on population increase and the lack of agricultural goods to feed the poor. Although there are neo-Malthusians who accept his projections feeling only that he erred in terms of perception of the agricultural revolution, Malthus on balance was wrong. 5

Jules Verne was correct in his prediction of space flights. And, in the 1920's Buckminster Fuller predicted the present environmental problem as well as the great technological leap which occurred after World War II. Much of De Toqueville's description of the U.S. appears to be germane. 10

Recently, forecasting in business, economics, the social sciences, and other areas has emerged. Scholars, with the help of the computer, have tended to veer away from concrete and specific predictions in favor of alternative models, parametric speculation, and projection.

Shonfield<sup>11</sup> has noted the emergence of/the field of "futurology", and though he is optimistic, he does raise some questions about the limits of the field. He importantly notes that futurists should avoid specificity.

In the last quarter of the century, the social, technical, and economic impact of the future has become significantly important. (The field of futurism appears to be growing into voluntary groups, research organizations, and has received sponsorship from federal as well as multi-national corporations). 12 Academic courses

dealing with the future are increasingly being offered. 13 McHale's 14 study of the field of futurism indicates nearly 3000 people involved in the area, with most occupied in free lance, corporate, or government work.

Due to the nature of the topic, both the theoretical assumptions and methodological approaches are diverse and varied. Some propose viewing the future in a Hegelian framework.

The Hegelian approach in futurism is not necessarily unique. Based on the premise of a conflict of forecasters, this approach deals with forecast, counterforecast, and revised forecast (the new synthesis). Its attributes are, included consensus and test of ideas, but it is flawed in the sense that the original forecast may be in error or that single particular elements are wrong. 15

The "zig-zag" or morphological design tries to avoid

"vertical" (Aristotelian, sequential) thinking. Rather,

lateral analysis is emphasized. Accordingly, one must not

develop a pattern, but rather restructure it. In vertical

thinking, one proceeds on a step-by-step process. In this

method, one avoids sequence. Its purpose is not so much to find an answer, but to provoke a new unstable structuring of the situation and let the latent possibilities of the development of new ideas occur. 16

Analogous systems is at its basis inductive logic. Extrapolation is based on the similarity of a present system and a proposed future model. At times the analogy design is called a railway system because it alludes to the fact that one system passes through or precedes like a train and another is following, but some distance back. The analogy suffers in the sense that the original model may be historically unique, the analogy loose, or that the future generations are aware of the previous outcome and can raise alternatives to change history. 17

Porecasting by analogy can involve the use of some past historical event or a well known process in order to predict a future occurrence. There must be sufficient resemblance in the two incidents being compared for the earlier one to be an accurate predictor of the future one. The method of analogy is most useful when the data required

for more precise methods is not available. Its most obvious difficulty is in determining whether or not the critical factors present in the original situation are likely to be present in the future.

Cyclical projections look for non-chance regularities over various time sequences. This approach assumes that the observed non-chance phenomenon can be recorded. Non-chance regularities can also individualize trends in terms of future change. 18

Dr. Olaf Helmes and Dr. Norman Dalkey of the RAND

Corporation developed the Delphi technique of forecasting
the future. The Delphi method involves the use of a panel
of experts. Each expert is asked his opinion on the future
developments in a particular field. The opinions are kept
anonymous. Each member of the panel is then given feedback
on the way other panel members replied to questions
and a second round of forecasting begins. This process is
repeated until a fairly consistent group forecast emerges.

Some of the advantages of this method are:

- The workloads for the experts on the panel and the coordinator are reasonable.
  - 2) It avoids the biases of a single expert.
- 3) The anonymity of the members avoids undue influence by a single person as biases produced by the interacting personalities in the group.
  Some of the disadvantages include:
- Excessive amounts of time spent in obtaining a consensus in this manner.
  - 2) It is expensive.
- 3) It depends on the competence of the panel members and ability of the administrator.<sup>19</sup>

Trend line merely assumes that current phenomenon will continue in the same degree as presently experienced. Thus, if the population is increasing an extra one per cent a year, future projection will continue on this assumption. Trend lines have been useful for short-range projections, but err on longer estimates because of societal modifications. 20

Envelope forecasting emphasizes the change of technology and its impact on a social system. It encourages one to think not only of change wrought by technology, but the synergism of many such changes. The envelope curve represents the values of the technical attribute over time, without regard to antecedent events. To the envelope curve, one adds numerous devices. From the total picture, one draws a smooth curve tangential to all the curves of the individual devices. This method indicates the general trend of technological innovations. <sup>21</sup>

Scenario writing assumes that large complex systems in terms of future system require descriptive analysis as opposed to qualitative ones. They generally are loosely structured overviews of alternative future systems. For specific elements, however, computer generated data is used. In some instances "science fiction" or "speculative fiction" writers are utilized in this method.<sup>22</sup>

Social-cultural forecasting involves a qualitative description of alternative future systems. It differs from scenario writing in the sense its emphasis is upon the economic infra structure and social organization.

In a sense it is a system approach which intertwines both a Marxian and functional analysis. 23

There are numerous other methods which are in many ways subsets of those listed above. One might want to refer to articles by de Jouvenel, Martino, and McHale. Research in the area indicates that socio-cultural forecasting, scenario building, Delphi technique, and trend lines seem to be the most popular. 25

These varying approaches, which include both their own theoretical assumptions and the methods to initiate them, may best be thought of as overlapping rather than mutually exclusive. Parenthetically, controversy appears to be brewing as to the general nature of the future. The controversy deals with the progression or integrative theory which promises that the future is generally beneficial to all, as opposed to the projection that ensuing social and economic malaise (such as doomsday or Walden theories) will occur.

Though the field of social and economic futurism is indeed immature and sometimes wrong, we feel there is justification in considering it an important discipline.

Years ago, the Hoover Commission successfully predicted certain aspects of later life styles in the United States. 26

As Ogburn has noted, social institutions are relatively slower to change than technical and physical structures. 27

This can give one some stability in terms of our future projection.

One must make projections broad enough to indicate limits, but not so wide as to be ultimate tautologies.

There is a fundamental common sense as well as logic in forecasting. As Bertrand de Jouvenal notes: "We have to make wagers about the future; we have no choice in the matter". 28

#### Chapter One

#### Footnotes

- l"Looking Toward the Twenty-first Century," America, January 11, 1969, p. 35.
- <sup>2</sup>Brownslaw Malinowski, <u>Magic, Science, and Religion</u> (New York: Doubleday Books, 1964).
- <sup>3</sup>Gerald Clarke, "Putting the Prophets in Their Place," <u>Time</u>, February 15, 1971, pp. 38-39.
- Alvin Toffler, <u>Future Shock</u> (New York: Bantam Books, 1970), p. 215.
- John K. Galbraith, Economics, Peace and Laughter, (Signet Books, 1971).
- <sup>5</sup>William Peterson, <u>World Population</u> (London: Macmillan & Co., 1969).
- Anton Skreja, "The Future Population of the United States," Population Bulletin, (April, 1971).
- 7Harison Brown, "Herman Cohn in Retrospect: A Parable," The Futurist, August, 1971, pp. 157-159.
- William J. Gue, "Jules Verne, The Prophet of the Space Age," <u>The Futurist</u>, April, 1971, pp. 76-78.
- 9Barry Farrell, "View from the Year 2000," Life, February 26, 1971, pp. 46-48.
- 10Andrew Shonfield, "Planning the Future," Current,
  March, 1969, pp. 41-53/

11 Ibid. pp. 41-53.

12"Institute for the Future Will Begin Operations in Connecticut This Fall," <u>The Futurist</u>, August, 1968, pp. 65-68.

Olaf Helmer, "New Attitudes Toward the Future," The Futurist, January, 1967, pp. 8-9.

Frank Snowden Hopkins, "Future Research: An International Effort," The Futurist, August, 1969, pp. 105-106.

Robert Rheinhold, "Rise of History of Science is a Reply to Technology," New York Times, February 18, 1970.

Erich Jantsch, "Technological Forecasting in Planning and Its Institutional Implications," Revista Espanola de la Opinion Publica, July-December, 1970, pp. 21-22.

See entire issue of Science Journal, October, 1967.

"Year 2000 Commission: 'Preposterous' Idea Bears Fruit," The Futurist, October, 1967, pp. 65-66.

Annette Harrison, <u>Bibliography on Automation and</u>
<u>Technological Change and Studies of the Future</u>, The Rand
Corporation, Santa Monica, 1971.

"What Future for Futurology," <u>UNESCO Courier</u>, April, 1971, pp. 4-32.

William H. Honan, "They Live in the Year 2000,"
New York Times, April 9, 1967, pp. 56-57.

Nigel Calder, "Future Research," New Statesman, September 29, 1971, pp. 399-400.

13 Ossip K. Fletcheim, et al, "Teaching the Future," History and Futurology, October, 1966.

"Professors Teach the Future," The Futurist, June, 1970, pp. 97-100.

14 John McHale, "A Survey of Futures Research in the United States," <u>The Futurist</u>, December, 1970, pp. 200-201.

<sup>15</sup>Frank L. Moreland, "Dialectical Method in Forecasting," <u>The Futurist</u>, August, 1971, p. 169.

Hans-George Gadamer, "Notes on Planning for the Future," Daedalus, Spring, 1966, pp. 572-589.

<sup>16</sup>Edward de Bono, "Zig Zag Thinking," <u>The Futurist</u>, February, 1970, pp. 29-31.

17Lt. Col. Joseph Martino, "Forecasting by Analogy," The Futurist, December, 1969, p. 162.

<sup>18</sup>Edward Dewey, "Forecasting by Means of Cycles," <u>The Futurist</u>, June, 1969, pp. 77-78.

<sup>19</sup>Joseph P. Martino, "Computers in Technological Forecasting," <u>The Futurist</u>, October, 1971, pp. 205-206.

Thomas J. Gordon, "Forecasters Turn to 'Delphi,"

The Futurist, February, 1967, pp. 1-2.

Joseph P. Martino, "The Consistency of Delphi Forecasts," The Futurist, April, 1970, pp. 63-69.

John P. Dory and Robert J. Lord, "Does 'TF' Really Work," <u>Harvard Business Review</u>, November-December, 1970, pp. 16-22.

Robert Ayres, <u>Technological Forecasting and Long</u>
Range Planning (New York: McGraw-Hill, 1969).

Erich Jantsch, <u>Technological Forecasting in Perspective</u>, Organization for Economic Cooperation and Development, Paris, 1967.

James Bright, <u>Technological Forecasting for Industry</u> and <u>Government</u> (Englewood Cliffs: Prentice-Hall, 1968).

Martin J. Cetron, <u>Technological Forecasting: A</u>
Practical Approach, (New York: Gordon & Breach, Science
Publishers, 1969).

Joseph Martino, "Evaluating Forecasts," The Futurist, June, 1969, p. 75.

Joseph Martino, "Forecasting for Research and Development," The Futurist, August, 1969, pp. 107-108.

Joseph Martino, "The Methodology of Technological Forecasting," The Futurist, April, 1968, p. 34.

A Bibliography of Selected Rand Publications, The Rand Corporation, Santa Monica, 1973.

20 Morris L. Earnst, <u>Utopia-1976</u>, (Rheinhart & Co., 1955).

Joseph Wood Krutch, "What the Year 2000 Won't Be Like," Saturday Review, January 20, 1968, pp. 12-14.

21John P. Dory and Robert J. Lord, "Does 'TF' Really Work?" Harvard Business Review (November-December, 1970), pp. 16-22.

<sup>22</sup>Lyman Sargent, "Utopia and Dystropia in Contemporary Science Fiction," <u>The Futurist</u>, June, 1972, pp. 93-99.

Kenneth F. Watt, "A Tragedy of Errors," Saturday Peview, November 25, 1972, pp. 56-57.

H. Bruce Franklin, "Fictions of the Future," The Futurist, February, 1970, pp. 26-28.

23Henry Winthrop, "The Sociologist and the Study of the Future," American Sociologist, May, 1968, pp. 136-145.

"The Next Five Centuries: A Prospective History Based on Current Trends," The Futurist, October, 1968, pp. 85-90.

"Commission on the Year 2000 Begins Work," The Futurist, January-February, 1967, p. 13.

Raymond Bauer, Social Indicators, (Cambridge: MIT Press, 1967).

John McHale, "Planning for the Future," Current, September, 1969, pp. 16-25.

John McHale, "Problems in Social and Cultural Forecasting," <u>Second International Future Research Conference</u>, April, 1971.

24 "The Art of Thinking About the Future," <u>The Futurist</u>, December, 1967.

25Barbara Hubbard, "An Appetite for Evolution," The Futurist, February, 1973, pp. 26-29.

Lyman Sargent, op.cit.

Mark R. Hillgas, The Future as a Nightmare: H.G. Wells at the Anti-Utopians (New York: Oxford University Press, 1967).

Burnham Beckwith, "The Future of Man: Optimism vs. Pessimism," The Futurist, April, 1972, pp. 62-66.

Wes Thomas and Jeremy Wiesen, "The Case for Optimism," The Futurist, April, 1972, pp. 67-69.

Barbara Hubbard, "Lancelot Law Whyte and the 'Integrators,' The Futurist, August, 1968, pp. 81-82.

Glen T. Seaborg, "The New Optimism" The Futurist, December, 1969, pp. 157-160.

Joseph Martino, "Studying the Future at the Armed Forces Staff College," The Futurist, August, 1968, pp. 81-82.

Barbara Hubbard, "Pollyannism in Reverse: A Threat to Man's Future," The Futurist, December, 1968, p. 131.

- 26william H. Honan, op.cit.
- <sup>27</sup>William F. Ogburn, <u>Social Change</u>, (New York; Viking Press, 1950).
- Gerald Clark, "Putting the Prophets in Their Places," Time, February 15, 1971, pp. 38-39.

Chapter Two

Social Implications

### Social Implications

## Introduction

This brief chapter summarizes the changes which will have impact upon your life-style during the next 50 years.

It will look at "super-industrialism," "post-industrialism," and the cities to and from which you will move.

### Early History

The beginning of social organization can be traced back to highly mobile, sparsely populated human groups which survived by hunting animals and gathering crops. Learly city life came after the domestication of animals and the movement towards an agrarian economy. Though the history of the city can be traced back to 3500 B.C., urban life as an American national phenomenon did not emerge until the end of the 1920's. The basic transition from early hunting and gathering units passed into agricultural communities and from there, with the assistance of industrialism, the city emerged. As Toffler notes,

the majority of the people in the world today live in agrarian settings. About 25% of the world, however, reside in an industrial milieu, and a small 1% to 3% now live in a super-industrial world.

### Super-Industrialism

People found in small enclaves in and around the megapolitan centers of the west and mid-Atlantic sea belt are thought to be residing now in tomorrow's world. 4 Much of their life is based on mobility, transience and rapid change. Industrial or super-industrialism as suggested by Bell, Boulding, and Toffler is a highly mobile world triggered by rapid travel, instant communication, and supported by a service sector economy. 5 The bases for their projections are generally evolutionary in character, and are for the most part based upon trend line assumptions.

According to these premises, what will the country look like some 50 years from now? Toffler sees mobility as a big factor that will continue to diminish culture differences among various regions in the country. If

he is correct, even the Plains region may have characteristics similar to the highly populated coastal areas. 6

However, rapid and even fluid movement of people is but one aspect of this widespread version of the future. Other changes include 1) higher productivity and improved efficiency of the industrial system, 7 2) less standard-ization and conformity with the rise of specialty subcultures, 8 3) continuance of a Keynsian market system, 9 4) less material scarcity for the occidental world and in particular the American social system, 10 5) rapid turn-over in media personalities, art forms, and language usage. 11

Beckwith in his <u>The Next 500 Years</u> envisions an industrial world with increases in specialization, professionalism, large-scale production, automation, and the growth of monopolies. He also indicates that education will continue to remain important along with scientific research. Population will increase along with "real" income. 12

Herman Kahn sees an increasingly sensate, bourgeois, centralized world which is more westernized as well as urbanized. 13 Esfandiary projects a world of increasing individual rights with world wide universalism. 14 A General Electric study indicates there will be increases in leisure, decentralization, rationality, and social justice. 15

As the future world appears to be increasing its urbanization (by the year 2020, around 80% to 90% of the population will live in urbanized areas), the farm appears to play a new role. The future farm will probably be large family corporations working huge biological monocultures. Land value will increase as farmers produce more and more food. There will be few small scale farmers. The farm population will dwindle from 3.5 million today to 1 million farmers or less. Strategic financial accounting, planning, and large air lifted machines will be used. Productions will be greatly increased.

### Post-Industrialism and Technology

In the technical world the changes are, however, more rapid. 20 Moore projects elaborate communication centers within the home. 21 Asimov envisions more personal air travel for the individual citizen. 22 Toffler sees ocean farming, weather modification, genetic control, new forms of reproduction, and elementary robots. 23

Kahn and Weinberg see these kinds of technological changes very likely to occur by 2000:24

Human hibernation for relatively extensive periods (months to years).

Improved capability to "change" the sex of children and/or adults.

Physically non-harmful methods of overindulging.

Chemical methods for improving memory and learning.

Simple techniques for extensive and "permanent" cosmetological changes (features, "figures", perhaps complexion and skin color, and even physique).

Stimulated and planned and perhaps programmed dreams.

Artificial moons and other methods for lighting large areas at night.

The most comprehensive list of technical changes was indicated by a Delphi forecast and it is listed here. 25

Scientific Breakthroughs Forecast by Delphi Experts

1970-1980:

Economically useful desalination of sea water.

Effective fertility control by oral contraceptive or other simple and inexpensive means.

Development of new synthetic materials for ultralight construction.

Automated language translators.

New organs through transplanting or prosthesis.

Reliable weather forecasts.

Operation of a central data storage facility with wide access for general or specialized information retrieval.

1980-1990:

Reformation of physical theory, eliminating confusion in quantum-relativity and simplifying particle theory.

Implanted artificial organs made of plastic and electronic components.

Widespread and socially widely accepted use of non-narcotic drugs (other than alcohol) for the purpose of producing specific changes in personality characteristics.

Stimulated emission (lasers) in X and Gamma-ray region of the spectrum.

Controlled thermo-nuclear power.

Creation of a primitive form of artificial life (at least in the form of self-replicating molecules).

Economically useful exploitation of the ocean bottom through mining (other than off-shore oil drilling).

#### 1990-2000:

Feasibility of limited weather control, in the sense of substantially affecting regional weather at acceptable cost.

Economic feasibility of commercial generation of synthetic protein for food.

Increase by an order of magnitude in the relative number of psychotic cases amenable to physical or chemical therapy.

Biochemical general immunization against bacterial and viral diseases.

#### 2000-2010:

Feasibility (not necessarily acceptance) of chemical control over some hereditary defects by medication of genes through molecular engineering.

Economically useful exploitation of the ocean through farming, with the effect of producing at least 20% of the world's food.

Biochemicals to stimulate growth of new organs and limbs.

#### 2010-2020:

Feasibility of using drugs to raise the level of intelligence (other than as dietary supplements and not in the sense of just temporarily raising the level of apperception).

Man-machine symbiosis, enabling man to extend his intelligence by direct electromechanical interaction between his brain and a computing machine.

Chemical control of the aging process, permitting extension of life span by 50 years.

Importantly in space, a lunar colony and more sophisticated space explorations may exist. 26

### Primary Institutions of the Future

Prognostications of the most basic human group, the family, indicate that it too will change somewhat. Though social institutions do not change as rapidly as techno-. logical applications, there will be some variance occurring in the family. Many indicate that the new-local family will change slowly but will continue to exist among the majority of the population. Women's status will improve with gradual sex role changes, improved job opportunities, and day care centers. The number of children per family will decline, sex of the child may be chosen, and there may be new forms of adoption. Young children may choose wider variations in the sex role with the basis of the role becoming unisexualism. Among a significant minority of the population, new non-bloodline "stem"

families and communal systems may emerge. 33 Importantly, mate choice may be more determined by continuity of personalities between individuals than sexual excitement or romantic love. 34

# Secondary Institutions of the Future

It is felt that voluntary associations will change in the future toward more "affinity" groups. 35 One institution of particular importance is the Organized Church. 36 There are many predictions which emphasize that religion will continue to be expressed in conventional ways and that there will be a considerable proportion of the population retaining traditional beliefs. 37 However, some changes appear to be emerging. First, the modernist churches (e.g., Presbyterian, Methodist) appear to be losing members and financial support while fundamental groups are gaining. Second, churches on the political left and humanist churches appear to be growing, even though their organizational strength is not as pronounced. 38 Some see a closer alliance between the social sciences and the Church, and increasing stresses in the

ministerial role.<sup>39</sup> Others have raised issues about the internal form of the Church.<sup>40</sup> As an overview, the future of religious systems appears to be polarizing around conservative and liberal churches. There is also a new emphasis of emerging small congregational-oriented churches which can re-emphasize primary ties.

Formal associations still appear to be growing and the large corporate organizations will still be the dominant form. 41 The static, conformity-oriented bureaucracy as defined by Whyte, is likely to give way to the new adhocracy as described by Toffler and Bennis. 42 As Bennis notes:

Replacing bureaucracy will be adaptive, problem solving, temporary systems of diverse specialists linked together by coordinating and task-evaluating executive specialists in an organic flux.

Such a flexible structure may enhance the satisfactions intrinsic to the task, but will reduce commitment to work groups. Job mobility will: increase. 43

Increasingly, employees will seek non-monetary rewards as work motivations and will have differing attitudes than in the past. 44 Managers will be able to

improve their efficiency through improved communication systems. 45 Forecasting, along with research and development will play an even more prominent role. 46

# Interactions and Life Styles

It appears that the class system will prevail in the United States. Though there will be social inequality, one source noted decreases in the number of poor. Others see the diminishment of poverty by an income floor. Dreikurs, a futurist psychiatrist sees major increases in social equality. For the middle classes, by 1985 the model household shall earn over \$10,000 (1971 dollars) to \$15,000, and by the year 2000, the average income will be between \$18,000 and \$20,000 (1971 dollars).

With the diminishment of poverty, racial tensions have the possibility of lessening. 54 However, most see some kind of accommodation but with continuing racial and ethnic separation. 55 And real harmony as such is not projected because of the historical struggle between groups, especially those with third world heritage.

Blacks because of physical visibility as well as entrance into the social system as slaves, may still find residual stigma upon them. <sup>56</sup> Chicanos, <sup>57</sup> as well as Indians, <sup>58</sup> or Neo-Americans, <sup>59</sup> may see a similar situation.

However, for the southern and eastern European community the picture is brighter. Increasingly, "cosmopolitans" will probably have improved positions in the American social system. 60

If one could give an overview of life styles in the super-industrial state in the year 2020, it is one of fluidity of relationships and geographic mobility. 61 Generally both consumption 62 and leisure should be on the increase. 63 There may even exist new and varied art forms. 64

Communications will be ever more efficient, cheaper, and more instant. 65 It is also thought that more forms of printed matter will be superceded by electronic communication devices. 66

However, in the midst of all this, man may find himself unable to cope with his feelings of aloneness, powerlessness, and alienation, a la "future shock". 67 To
overcome this he may seek out religion, psychiatry, 68
pharmacology, 69 or group therapy. 70 For the individual

of the future, struggle for material goods may be supplanted by quest for self.  $^{71}$ 

### Social Implications of Land Use

How does the super industrial life style translate into land use and land formation? What will the future city look like because of social implications?

John McHale has emphasized the importance of life style upon future urban form. However, the field of social ecology traces back to Robert Park. Park likened man-to-land relationships to the biological world of ecology, and he coined the term "Social Ecology." Park and his Chicago School studied the windy city, and noted certain urban formations emerging from social relation—ships. Others have followed including Hoyt, Harris, and Ullman, as well as Skevky, Bell, and Williams. Among urban geographers, social ecology is called "internal form" or social space. 74

What general impact does super-industrialism have upon this area in terms of land use? Freedman and Miller note:

Looking ahead to the next generation, we foresee a new scale of urban living that will extend far beyond existing metropolitan cores and penetrate deeply into the periphery. Relations of dominance and dependency will be transcended. The older established centers, together with the intermetropolitan peripheries that envelope them, will constitute the new ecological unit of America's postindustrial society that will replace traditional concepts of the city and metropolis. This basic element of the emerging spatial order we shall call the "urban field".75

If social projections about super-industrialism hold, most cities will expand horizontally. The life style of high geographic mobility and space project urban expansion or scatteration. Increased leisure and fluidity of interaction generally encourage geographic distance and larger amounts of land consumption. As Freedman and Miller note, super-industrialism requires a wider life and land space, wider choice of living environments, and wider community of interests. 76

This projected life style also places new definitions upon the rural-urban fringe as well as difficulties in defining the urbanized area. 77

With this predicted consumption pattern the "centripetal pull" of the core of the city weakens because of
additional real income, more leisure time, and increasing
mobility. 78

Both "new towns" and suburban developments with inclusive resource endowments (retail for consumers, and manufacturing for workers) also weaken the "pull". 79

Super-industrialism engenders megapolitan growth.

Megalopolis of grows not only from natural increase but from migratory pressures of groups both from core and rural areas. "Urbs" and suburbs grow at the expense of both heartland and hinterland. This polarized development robs both core and rural areas of manpower and tax base. 81

What about the interior of the city and social formations within it?

First, if the neo-local family continues as the majority (but not the only preference) there will be a continuance of "life cycle segregation". What this means is that as one progresses through life, various regions of the city take on importance. 82 Generally, families take on zonal characteristics.

The concentric zone is a ringlike feature with Zone I, the core, being downtown. Zone II is a zone

of high land value, but low rent and low income as proprietors wait hopefully for core expansion so that they
can sell their land. Zone III is generally populated by
the working classes. Zone IV is populated by upper middle
income groups, inestablished suburbs, and Zone V is a
commuter zone composed of young lower-middle-class families.

The zonal approach is but a classic model with certain flaws, but evidence indicates that family patterns are based on zonal features. 83 Generally suburban residents are at peak income with no or older children. Inhabitants of the inner city are either young, single, or yet to achieve their maximum income. They may also be those who never quite achieve the necessary income for the suburbs or are large poor families. Using this model, family size rises and falls along the various rings. Zone. I is generally devoid of families. Zone II is heavily populated with large poor families, single, and older couples. Zone III has large families. Zone IV has smaller families with older children. Zone V has small young families.

Zonal formation spreads with the sprawl. If superindustrialism projection about family occurs in a similar
and correlative fashion, zonal formations will encourage
spread, for single nuclear families push outward looking
for single dwelling units.

If traditional class differences continue as suggested, this too will encourage scatteration. Hall has noted that in Europe, classes may reside close together because a social psychological distance exists. However, in the United States, a classless ethic is thought to be traditional form. Therefore, personal avoidance is difficult. For Americans, geographic distance then takes on an importance. Banfield notes that classes have difficulty mixing because when some new group of upper income people move into an area, their land improvements raise property tax, driving the poor. Generally, the poor, when they enter a more affluent area, drive out the middle and upper income by their life style.

The American class system also encourages spread.

Hoyt, who looked at rent data, calculated a sector theory

of urban life. This system sees the city divided by major

corridors forming sectors. Urban geographers have noted that class systems follow sectoral patterns. 87 On density-distance flow charts measured from the core of the city, geographers have noted the income crest along the traverse intersect of the inner suburbs. However, class may vary along the axial growth. Data shows stronger continuities within the sector than between them. 88

The traditional racial and ethnic separation also gives rise to sprawl. Generally, man and varied forces keep minority members impacted in restricted areas until they "spill over" into the ensuing neighborhoods.

Nationwide projections indicate that, at current rates, the black population should increase significantly. 89 With projection of the continuance of the ghetto, we see necessary expansion of it. 90 This will, in an interdependent way, increase the white European flight to even more newly developed suburban tracts.

Both Harris and Ullman as well as Shevky, Bell, and Williams, noted the creation of nucleated formations around urban neighborhoods. 91 Data of urban geographers

indicate that these neighborhoods flow along racial and ethnic lines.

Lastly, if time for leisure activity increases, new outdoor recreational areas will be needed. Generally in the past, parks have been difficult to establish because of the competition with developers for land. This is a common problem in urban expansion.

### Conclusion

If the super-industrial trends continue, metropolitan areas will probably continue to expand horizontally. Zonal life cycle patterns, sectoral class divisions, and nucleated ethnic formations will continue to expand. More leisure areas will be demanded.

### Chapter Two

#### Footnotes

<sup>1</sup>Bronislaw Malinowski, <u>Magic, Science</u>, and <u>Religion</u> (New York: Doubleday Anchor, 1948).

Leonard Broom and Philip Selznick, Sociology - A
Text with Adapted Readings (New York: Harper & Row, 1973).

Society Today (DeMar, California: CRM Books, 1971).

George Lungberg, Clarence Schrag, Otto Larsen, and William Catton, Jr., Sociology (New York: Harper & Row, 1968).

Scott Greer, Social Organizations (New York: Random House, 1968).

George C. Homans, The Human Group (New York: Harcourt, Brace & World, 1950).

<sup>2</sup>Henry S. Churchill, <u>The City is the People</u> (New York: W.W. Norton, 1962).

John Sirjamaki, The Sociology of Cities (New York: Random House, 1964).

Vanishing Peoples of the Earth (Washington, D.C.: National Geographic Society, 1971).

3Alvin Toffler, <u>Future Shock</u> (New York: Random House, 1970), p. 56.

<sup>4</sup>Jesse M. Unruh, "To Bend the Wind," The Futurist, October, 1970, pp. 175-181.

"California: Turned-On Tomorrowland," The Futurist, April, 1967, pp. 19-20.

Billy Rojas, "Laboratories of the Future," The Futurist, October, 1971, p. 208.

5Kenneth E. Boulding, A Primer on Social Dynamics (New York: Free Press, 1970).

<sup>6</sup>Toffler, op.cit., p. 92.

Robert C. Turner, "The Long Future of the Economy," Business Horizons, Vol XIII, No. 3, June, 1970, pp. 5-12.

John D. Huss and Belle L. Negrin, "A Changing Society," The Futurist, August, 1969, p. 102.

8Toffler, op.cit., pp. 262-276.

George H. Brown, "1984. . . Plus 1," The Conference Board Record, Vol. VII, No. 12, December, 1970, pp. 20-24.

10Daniel Bell, "The End of Scarcity?" Society, April, 1972, pp. 49-52.

11 Alvin Toffler, "Rapid Turnover in Language," The Future Shock (New York: Random House), p. 171.

<sup>12</sup>Burnham P. Beckwith, "Beckwith's 31 Major General Social Trends," <u>The Futurist</u>, October, 1968, pp. 88-89.

13"Things to Come," The Futurist, February, 1973, pp. 8-15.

Herman Kahn, "Future: Toward the Year 2000," Wilson Library Bulletin, 47:251-3, November, 1972.

14 Wes Thomas and Jeremy Wiesen, "The Case for Optimism," The Futurist, April, 1972, pp. 67-69. <sup>15</sup>Ian H. Wilson, "How Our Values Are Changing," <u>The Futurist</u>, February, 1970, pp. 518.

Henry Clark, "The Changing Character of American Values," The Futurist, February, 1970, pp. 9-10.

Victor Ferkiss, "Toward the Creation of Technological Man," The Futurist, February, 1970, pp. 11-12.

Frank Snowden Hopkins, "The Rise of Technological Man," (Review) The Futurist, February, 1970, pp. 12-13.

Isaac Asimov, I, Robot (New York: Signet, 1950).

Daniel Bell (ed.), The Year 2000: Work in Progress (Boston: Beacon Press, 1967).

Peter F. Drucker, The Age of Discontinuity: Guidelines to Our Changing Society (New York: Harper & Row, 1968).

Gerald W. Smith, Society Today -- Resource Letter, "Social Change," Series 1.

16Interview with Martin Osterloh, farmer of 14 years, with 450 acres in northeast Nebraska.

Interview with Ken Hagen, northwest Ohio grain farmer, 450 acres.

"Saving the Farm -- and the City," New York Times, September 5, 1972.

Orville L. Freeman, "American Life in the Year 2000," The Futurist, October, 1967, pp. 76-77.

Irwin P. Halpern, "Growth Centers and New Communities," The Futurist, October, 1970, pp. 163-164.

- Clifford M. Hardin, "Steps to Improve Life in Rural Areas," The Futurist, October, 1970, pp. 165-166.
- 17 "The Changing Profile of Indiana Dairy Plants,"

  Research Progress Report, Indiana Agricultural Experiment
  Station, Purdue University, Lafayette, Indiana, No. 358.
- T. Kristensen, "The National Economy and Farming in Swiftly Changing World--Directions and Understanding," Agriculture and Agrapolitics, University of Denmark, 1969.

Bram Peper, "Agricultural Policy and Social Policy," Rural Sociology, 9:3:221-234, 1969.

- <sup>18</sup>P.L. Arcus, "Projections of Farm Employment in the United States through 1980," (Dissertation Abstract), 1917, Ann Arbor, Michigan.
- G.L. Williams, "Projecting Manpower Requirements and the Use of Educated Manpower in Agriculture," <u>Mediterranea</u>, 28:762-73, Paris, 1968.
- 19L.V. Mayer, "An Analysis of Future Resource Supplies, Resource Utilization, Domestic and Export Demand, and Structural Change in the Agricultural Economy to 1980," (Dissertation Abstract) 28:4:1173-1174, Ann Arbor, Michigan, 1967 (Order No. 67-12, 979).
- I. Hock and N. Tryphonopoulos, "A Study of the Economy of Napa County, California, " Giannini Foundation Resource Report, California Agriculture Experimental Station, No. 303, Berkeley, California, 1969.
- C.P. Butler, "Projecting Economic Growth Potentials of Dairy Farms in South Carolina," No. AE-318, Southern Carolina Agricultural Experimental Station, Clemson, South Carolina, 1968.

- W.E. Saupe and D.R. Kaldor, "Efficient Organization of the Farm Industry in the NorthCentral Region of the United States in 1959 and 1980," Research Bulletin, Agricultural Home Economic Experimental Station, Iowa State University of Science Technology, Amer, Iowa, 560: 153, 1968.
- M. Kohne, "Objectives, Problems and Possibilities of Long Range Assessment of the Future Agricultural Supply Situation," 2:241-67, Hamburg station 47, 1969.
- R.H. Tremblay and C.W. Bigalow, "Projections of Dairy Farm Numbers in Vermont," <u>Bulletin</u>, Vermont Agricultural Experiment Station, Burlington, Vermont, 1968.
- C.S. Hunt, G.W. Olson, and E.E. Hardy, "The Oswego River Basin: Projections of Agricultural Production," <u>A.E. Research No. 276</u>, New York State Agricultural Experimental Station, Cornell University, Ithaca, New York, 1969.
- G.W. Dean and B.A. King, "Projection of California Agriculture to 1980 and 2000: Potential Impact of San Joaquin Valley West Side Development," Giannini Foundation Resource Report of California, No. 312, Agricultural Experiment Station, Berkeley, California, 1970.
- H. Patzold and W. Stottmeister, "Keeping Cows Indoors All the Year versus Letting Out to Grass During the Summer," University of Berlin, Berlin, 2:213-8, 1968.

20 Toffler, op.cit., p. 26.

Herman Kahn, "The Year 2000," (reprinted) Wilson Library Bulletin, 47:251-3, November, 1972.

Richard Kostelanetz, The Edge of Adaptation: Man and the Emerging Society, (Spectrum Books, 1973).

J.R. Moskin, "Year 2018," Look, 32:110, May 28, 1968.

Herman Kahn, "Prospects for Mankind: 1990," Saturday Evening Review, 244:52-53.

G.R. Salancik, "Choosing Technology for a Better World," The Futurist, October, 1972, pp. 208-209.

Robert C. Turner, "The Long Future of the Economy,"

<u>Business Horizons</u>, Vol. XIII, No. 3, Indiana University

Graduate School of Business, June, 1970, pp. 5112.

Dennis Livingston, "Science Fiction and Futurology: Some Observations at a Science Fiction Convention," The Futurist, June, 1968, pp. 47-48.

Frank Snowden Hopkins, "Stuart Chase Envisions the 'Most Probable' World," The Futurist, August, 1968, p. 82.

Joseph P. Martino, "The Pace of Technological Change," The Futurist, April, 1972, pp. 70-72.

21J. Jamison Moore, "What Business Should Do Now to Anticipate the Future," <u>The Futurist</u>, February, 1969.

22"New Book Assembles Asimov's Forecasts," The Futurist, April, 1969, pp. 49-50.

23Toffler, op.cit., pp. 190, 192, 197-198, 200,205,
and 212.

24 "The Next 32 Years as Seen by Herman Kahn and the Hudson Institute," The Futurist, February, 1968, pp. 1-4.

"The Next Five Centuries: A Prospective History Based on Current Trends," <u>The Futurist</u>, October, 1968, Pp. 85-87.

25 "Developments in Space Forecast by Delphi Experts," The Futurist, December, 1968, p. 129. <sup>26</sup>"Lunar Colony for People of All Nations?" <u>The Futurist</u>, February, 1972, p. 12.

Mary Taylor Coates, "Manned Space Stations: Are They Worth the Price?" The Futurist, April, 1973.

Arthur C. Clarke, "Satellites and the United States of Earth'," The Futurist, April, 1972, p. 61.

"Corporation Issues Major Forecast of Future Technology," The Futurist, October, 1967, pp. 68-69.

Ralph E. Hamil, "Moving Beyond the Earth," The Futurist, April, 1971, pp. 68-71.

"What Lies Ahead in Space?" The Futurist, April, 1971, pp. 61-64.

"Exploring the Universe," (NASA Report) The Futurist, April, 1971, pp. 65-67.

Frederick C. Durant, III, "Conquest of Space is Viewed as Man's Next Evolutionary Step," The Futurist, October, 1968, pp. 97-98.

Earl Hubbard, "The Need for New Worlds," The Futurist, October, 1969, pp. 117-118.

Richard A. Givens, "New Aspirations for Man in the Space Age," The Futurist, October, 1969, pp. 118-119.

"Man's Future in Space," The Futurist, October, 1969, p. 120.

Wes Thomas and Jeremy Wiesen, "The Case for Optimism," The Futurist, April, 1972, pp. 67-68.

27W.F. Ogburn, <u>Social Change</u> (New York: Viking Press, 1950). <sup>28</sup>James A. Peterson, "Marriage Becomes More Popular as Divorce Rate Rises," <u>The Humanist</u>, May-June, 1970.

G.C. Gallen (Ed.) <u>Indian Journal of Social Research</u>, Baraut, India.

"The Future of the Family," Social Science, 42:3: 168-175, June, 1967.

John N. Edwards, "The Future of the Family Revisited," Journal of Marriage and the Family, 29:3:505-511, August, 1967.

Albert Ellis and David Mace, "The Use of Sex in Human Life," <u>Journal of Sex Research</u> (Institution of Rational Living, New York, and American Association of Marriage Counselors, Madison, New Jersey) 5:1:41-49, February, 1969.

H.S. Resnick, "How Shall We Raise Our Children in Year 2000," <u>Saturday Review Education</u>, March, 1973.

<sup>29</sup>Margaret Cussler, "The Future of the Role of the Woman Executive," <u>A Sociological Abstract</u>, F5586-sss, no. 0380, 1972.

Simone de Beavior, The Second Sex (New York: Bantum Books, 1965).

Suzanne Keller, "Does the Family Have a Future?"

(A Warner Modular Publication) Journal of Comparative
Family Studies, Spring, 1971.

Leo Davids, "Personal Roles: Sex, Ethnic, and Family," (Section World Future Society, First General Assemble, May 14, 1971).

Herbert A. Otto (Ed.), Family in Search of a Future (New York: Appleton-Century-Crofts, 1971).

John B. Parrish and Jean S. Block, "The Future of Women in Science and Engineering," <u>Bulletin of the Atomic Scientists</u>, 24:5:46-49, May, 1968.

Nona Glazer-Malbin and Helen Youngelson Waehrer, Woman in a Man-Made World (Chicago: Rand McNally & Co., 1972).

Eleanor Garst, "Will the Family Become Obsolete," The Futurist, December, 1967, p. 95.

Betty Friedan, The Feminine Mystique (Dell Publishing, 1963).

Jeanne Binstock, "Motherhood: An Occupation Facing Decline," The Futurist, June, 1972, pp. 99-102.

Jessie Bernard, "Women, Marriage, and the Future," The Futurist, April, 1970, pp. 41-42.

Sally Woodhull Cornish, "Marriage has Many Futures," The Futurist, June, 1972, p. 103.

"Latest Rules on Jobs for Women," U.S. News and World Report, 68-87, June 22, 1970.

Nancy Grouchow, "Discrimination: Women Charge Universities, Colleges, with Bias," Science, 168:559-61, May 1, 1970.

30 Suzanne Keller and Siegwart Lindberg, "When Parents Can Choose--Which Sex Will It Be?" The Futurist, October, 1972, pp. 193-196.

"Child Market," Newsweek, June 16, 1969, p. 70.

- 31 Marshall McLuhan and George B. Leonard, "The Future of Sex," Look, July 25, 1967, pp. 56-60.
- 32 Masturbation is Viewed with Increasing Favor by Physicians, Siecus, New York.

Albert Ellix, "Sexual Promiscuity in America," The Annals of the American Academy of Political and Social Science, June, 1968.

"Abortion Law Attack Gains," The Washington Post, November 11, 1969.

"Abortions: Moves to Abolish all Legal Restraints,"
The New York Times.

"Compulsory Childbirth," The Washington Post, November 18, 1969.

Tom Burke, "The New Homosexuality," Esquire, December, 1969.

"The Homosexual: Newly Visible, Newly Understood," Time, October 31, 1969.

William Parker, Homosexuality: A Selective Bibliography of over 3,000 items (Reviewed by Edward R. Sagarin), (Matuchen, New Jersey: Scarecrow Press, 1971).

Martin S. Weinberg and Alan B. Bell, Homosexuality:
An Annotated Bibliography (New York: Harper & Row, 1970).

Ralph W. Weltge, (Ed.), The Same Sex: An Appraisal of Homosexuality (Philadelphia: Pilgrim Press, 1969).

W. Dwight Oberholtzer, Is Gay Good? Ethics, Theology and Homosexuality (Philadelphia: Westminster Press, 1971).

Joseph A. McCaffrey (Ed.), The Homosexual Dislectic, (Englewood Cliffs, New Jersey: Prentice-Hall, 1972).

Male: Treatment for Men Troubled by Homosexuality (New York: McGraw-Hill Book Co., 1970).

M.P. Feldman and M.J. MacCulloch, <u>Homosexual Behavior</u>:

<u>Therapy and Assessment</u> (Oxford and New York: Pergamon Press, 1971).

George Weinberg, Society and the Healthy Homosexual (New York: St. Martin's Press, 1972).

Wainwright Churchill, Homosexual Behavior Among Males: A Cross-Cultural and Cross-Species Investigation (Englewood Cliffs, New Jersey: Prentice-Hall, 1971).

Colin J. Williams and Martin S. Weinberg, <u>Homosexuals</u> and the Military: A Study of Less Than Honorable Discharge (New York: Harper & Row, 1971).

Arno Karlen, <u>Sexuality and Homosexuality: A New View</u> (New York: W.W. Norton, 1971).

"The Good Guys, the Bad Guys, and the Gay Guys,"
(reviewed a Survey Essay in Contemporary Sociology, American Sociological Association, Vol. 2, No. 1, January, 1973, pp. 3-22.

33Keith Melville, "Communes in the Counter Culture," (New York: William Morrow & Co., Inc., 1972).

"Building Utopia in Virginia," The Futurist, December, 1967, pp. 86-87.

"Communes Become Popular," Alternatives Foundation, No. 1, Berkeley, California, Fall, 1969.

Kenneth Westhues, "Hippiedom - 1970; Some Tentative Hypotheses," Sociological Quarterly, 13:81-89, Winter, 1972.

Roy Ald, Youth Communes (Sower Publication, 1970).

B.F. Skinner, Walden Two (New York: MacMillan Co., 1972).

Stephen Diamond, What the Trees Said, Life on A New Age Farm (New York: Delta Book, Dell Publication, 1971).

Willis E. McNelly and Leon E. Stover (eds.), Above the Human Landscape: An Anthology of Social Science Fiction (Pacific Palisades: Good Year Publication, 1972).

Raymond Mungo, Famous Long Ago--My Life & Hard Times with Liberation News Service (New York: Beacon Press, Pocket Books, 1971).

William T. Gay, "Analysis of the Dual Family," The Futurist, April, 1969, p. 52.

Colette Carisse, "Family Values of Innovative Women: Perspective for the Future," <u>Sociological Abstract</u>, E 3780, ISA, 0060, 1970.

"The Connubial Utopias of Robert Rimmer," The Futurist, October, 1968, pp. 99-100.

Elsie Boulding, "The Family as an Agent of Social Change," The Futurist, October, 1972, pp. 186-190.

Larry and Joan Constantine, "Where is Marriage Going?" The Futurist, April, 1970, p. 44.

Roland Huntford, "Sweden Ponders New Marriage Legislation," The Washington Post, February 22, 1970.

Karl E. Meyer, "Sweden Redefines the Family," The Washington Post; April 13, 1969.

"From Sweden and Denmark: Experiments in Marriage," Life, August 15, 1969. Myron Orleans and Florence Wolfson, "The Future of the Family," The Futurist, April, 1970, pp. 48-49.

Jessie Bernard, "Bruno Bettelheim Thinks Child Rearing May Increasingly be Handled by Experts," The Futurist, April, 1970, p. 51.

Victor J. Papenek, "The Volita Project," The Futurist, April, 1970, pp. 52-58.

Helen M. Hacher, "How Clergymen View Hippiedom," Christian Century, 87:887-891, July 22, 1970.

- "The Future of the Family," Society Today (Palo Alto: CRM Books, 1971), p. 268.
- <sup>35</sup>Joel C. Snell, "Demographic Characteristics of Student Leaders in Instrumental and Expressive Voluntary Association," by Joel C. Snell and W.O. Wakefield, <u>The</u> <u>Southern Journal of Educational Research</u>, 5:3:159-168, July, 1971.
- 36Betty R. Scharf, <u>Sociological Study of Religion</u> (New York: Harper Torch Books, 1972).
- Harvey Cox, "The Future of Christianity and the Church," The Futurist, August, 1970, pp. 122-123.
- J. Edward Carothers, "What's Ahead for the Church?" and "The Church in the Year 2000," World Outlook, April, 1969.
- John C. Haughey, "Christians in America," America, January 10, 1970.

"Conference on Religion and the Future Impresses
Theologians with Rapidity of Change," The Futurist, August,
1970, p. 129.

T. J. Gordon, "Some Possible Futures of American Religion," The Futurist, August, 1970, pp. 131-132.

"The Church in the Future," Commonweal, October, 1969, pp. 133-155.

<sup>38</sup>Interview with Delvin D. Hutton, Associate Professor, Greek and Religion, Dana College, Blair, Nebraska.

Comments are also derived from one Jewish Rabbi, one Fundamentalist minister, and one Modernist theologian.

"The Psychologist, the Computer, and the University,"
(More College Students Profess no Religion), Washington
Star, August, 17, 1970.

Alvin Toffler, The Futurist, October, 1971, p. 194.

Joel C. Snell and William O. Wakefield, "Plains State Lutheran Ministers: Structural and Perceptual Sources of Alienation," (review of 40 some references in this area), The Iliff Review, XXIX, 2:25-34, Spring, 1972.

"The Twenty-First Century Demands New Forms of Ministry," The Futurist, August, 1970, pp. 135-136.

James F. Crow, "How Science is Reshaping Man's Views of Right and Wrong," The Futurist, June, 1968, pp. 45-46.

Neil P. Hurley, "The Church's View of a New World," America, January 11, 1969, p. 44.

Walter Wagner, "Future Morality," The Futurist, October, 1971, pp. 197-199.

40 "Church in the Year 2000: Commonweal Paper, Symposium," Commonweal, October 31, 1969, pp. 116-160.

- Anton Zandor Lavey, The Satanic Bible (New York: Avon Publishers, 1969).
- "Will 'Astro-Theology' be Needed Soon?" Time, January 24, 1969.
  - "Churchless Churches Forming," Time, April, 12, 1968.
- "Churches' Congregations Growing," Washington Post, May 31, 1969.
- "Will 'Inner Voyages' be Theology's Next Step?" Time, December 26, 1969.
- George H. Dunne, "Development -- A Christian Concern?"

  America, December 2, 1972, pp. 466-467.
- 41 Robert C. Turner, "The Long Future of the Economy,"
  Business Horizons, XIII: 3:5-12 (June, 1970).
- 42William H. White, Jr., The Organization Man (New York: Simon and Schuster, 1956).
- Warren G. Bennis, "The Coming Death of Bureaucracy;" Think (IBM Magazine), November-December, 1966.
  - 43 Ibid.
- 44 "Mapping Management to the 21st Century," Business Week, September 2, 1967, pp. 112-114.
- "The Growing Importance of Strategic Personnel Planning," <u>Business Management</u>, February, 1968.
- George Kozmetsky, "Reflections of a 21st Century Manager," The Futurist, June, 1969, pp. 73-74.
  - "The Next 65 Years, " Financial World, October 25, 1967.

- 46 Joseph P. Martino, "But Haven't I Been Doing that All Along?" The Futurist, June, 1972, pp. 120-121.
- Reinhard Bendix and Seymour Lipset, Class Status and Power -- Social Stratification and Comparative Perspective (New York: Free Press, 1966).
- 48"Number of Poor Declines", Poverty in the United States: 1959 to 1968 (Washington, D.C.: U.S. Government Printing Office).
- 49 Theobald's Alternative Future of America, " The Futurist, February, 1969, p. 22.
- "Preface: Government Policy and the Poor: Past, Present, and Future," <u>Journal of Social Issues</u>, 23:3:1-10, Summer, 1970.
- Rudolph Dreikurs, "Equality: The Life-Style of Tomorrow," The Futurist, August, 1972, pp. 153-157.
- 51"A Look at the Great Economy of 1985," Business Week, December 18, 1971, pp. 84-85.
- 52 George H. Brown, " 1984 . . . Plus 1, " The Conference Board Record, VII, 12:20-24, December, 1970.
  - 53 Turner, "The Long Future of the Economy," op. cit.
- Thomas F. Pettigrew, <u>Racially Separate or Together?</u>
  (New York: McGraw-Hill, 1971).
- 55 Nathan Wright, Jr. (ed.), What Black Politicians are Saying, (New York: Hawthorn, 1972).
- "Rising Clamour for Black Separatism," U.S. News & World Report, 69:82-83, September 21, 1970.

John Brigham and Theodore Weissbach, (eds.), Racial Attitudes in America -- Analysis & Findings of Social Psychology (New York: Harper & Row, 1972).

56 August Meier and Elliott Rudwick, From Plantation to Ghetto (New York: Hill & Long, 1970).

Milton Meltzer (ed.), <u>In Their Own Words -- A History</u>
of the American Negro (New York: Thomas Y. Crowell Co.,
1964).

A History of the National Urban League (Boston: Little, Brown, & Co., 1971).

Stephen J. Herzog, Minority Group Politics: A Reader (Chicago: Holt, Rinehart & Winston, 1971).

Melvin Drummer (ed.), <u>Black History -- A Reappraisal</u> (New York: Doubleday, 1968).

Andrew Billingsley, Black Families in White America (Englewood Cliffs: Prentice-Hall, 1968).

The Negro and the City, Fortune Book, Time, Inc., 1968.

Carl T. Rowan, "How Racists Use 'Science' to Degrade Black People," Ebony, 25:31-40, May, 1960.

"Black Policy-Makers: An Edge on Nixon," Life, 61-62, July 24, 1970.

Richard Rashbe, "Trust for Black Catholics?" Commonweal, 92:35-37, March, 1970.

"Who's Irresponsible?" New Republic, 162:10, May 9, 1970.

Steven Roberts, "Black Studies: More than 'Soul Courses', "Commonweal, 91:478-479, January 30, 1970.

Claude Brown, Manchild in the Promised Land (New York: The New American Library, Inc., 1966).

Kenneth B. Clark, <u>Dark Ghetto</u> (Scranton: Harper & Row, 1965).

Gerald Leinwand, The Negro in the City (New York: Simon & Schuster, 1968).

Elliott Liebow, <u>Tally's Corner</u> (Boston: Little, Brown & Co., 1967).

Malcolm X and A. Haley, The Autobiography of Malcolm X (New York: Grove Press, 1965).

August Meier and Elliott M. Rudwick (eds.), The Making of Black America (New York: Atheneum Pub., 1969).

Piri Thomas, <u>Down These Mean Streets</u> (New York: The New American Library, Inc., 1967).

"Black America 1970," Time, April 6, 1970.

U.S. Riot Commission Report: Report of the National Advisory Commission on Civil Disorders (New York: Bantum Books, Inc., 1968).

Louis Wirth, The Ghetto (Chicago: University of Chicago Press, 1928).

57 John H. Bruma, Mexican-Americans in the U.S. -- A Reader (Schenkman Publishers Co., Inc., 1970).

Joan W. Moore, <u>Mexican Americans</u> (Englewood Cliffs: Prentice-Hall, 1970).

1

"Changing Times: Chicanos-General," Nation, 209:460, November 3, 1970.

Don Post, "Mexican-Americans and 'La Raza'," Christian Century, 86:325-328, March 5, 1969.

Toni De Gerez, Three Times Lonely: The Plight of the Mexican Child in the American Southwest, (Horn Book, 46: 66-73, February, 1970).

58 Edgar S. Cahn (ed.), Our Brothers Keeper: The Indian in White America (New York: New Community Press, 1970).

<sup>59</sup>Harold E. Fey, "America's Most Oppressed Minority," The Christian Century, 88:65-68, January 6, 1971.

Rey Mickinock, "The Plight of the Native American," The Library Journal, 96:2848-2851, September 15, 1971.

Robert J. Trotter, "Indians and Sociologists: Science or Exploitation?" Science News, 100:234, October 2, 1971.

Anna Lee Stensland, "American Indian Culture: Promises Problems, and Possibilities," <u>English Journal</u>, 60:1195-1200, December, 1971.

"An Indian in the City: The Effects of Relocation," Newsweek, 77:94, June 14, 1971.

"Education Funds for Indians 'Misused' Charges NAACP,"
The Library Journal, 96:1413-1414, April 15, 1971.

Roul Tunly, "Smooth Path at Rough Rock," American Education, 7:15-20, March, 1971.

Paul Conklin, "Good Day at Rough Rock," American Education, 3:4-9, February, 1967. "American History: A Native American View," The Library Journal, 96:678, February 15, 1971.

"Making it on Your Own, Navajo Style," Nation's Business, 50:19 May, 1971.

Terri Schulz, "Bamboogle Me not at Wounded Knee," Harper's, 246:45-56, June, 1973.

Richard Boutwell, William C. Low, Kristin Williams, and Thomas Proffit, "Red Apples," <u>Journal of American Indian Education</u>, 12:11-14, January, 1973.

Robert Ritzenthales, "Cultural Innovation," <u>Journal</u> of American Indian Education, 11:20-22, May, 1972.

"Search for Identity Creates Problems for Indian Students," The Journal of American Indian Education, 10:7-10, October, 1971.

The American Indian Reader -- Education, (The Indian Historical Press, Inc., American Educational Publishers, San Francisco, California, 1972).

Roy Bongartz, "Who am I? The Indian Sickness: The White Hank Case," Nation, 210:496-498, April 27, 1970.

60 Peter Schrag, The Decline of the Wasp (New York: Simon & Schuster, 1970).

Peter Binzen, Whitetown USA (New York: Vintage Books, 1970).

Edgar Litt, Ethnic Politics in America (Scott, Foresman & Co., 1970).

"Italian Power," Newsweek, 75:22, June 22, 1970.

Michael Novak, The Rise of the Unmeltable Ethnics (New York: MacMillan, 1971).

- Andrew M. Greeley, Why Can't They Be Like Us? (New York: Dutton, 1971).
  - 61 Toffler, op.cit., p. 155, 158, 166.
- Carl H. Madden, "Business and the Future," The Futurist, December, 1969.

Rudolph Klein, "Growth and Its Enemies," Commentary, June, 1972.

Robert C. Turner, "The Long Future of the Economy," op.cit.

Kenneth E. Goulding, "Prognostics: A Guide to Present Action," Saturday Review, February 10, 1968, pp. 36-37.

Donald Michael, "An Ironic and Tentative World," The Futurist, April, 1967, pp. 21-22.

63 Arthur Roberts, "Vision of the Future," Bulletin of the Atomic Scientists, May, 1969.

William R. Ewald, Jr. (ed.), <u>Environment and Change:</u>
The Next Fifty Years (Bloomington: Indiana University
Press, 1968).

"Vacationing -- Houseboat Style," Tomorrow, January,

"Kicking the Habit," Newsweek, March 2, 1970.

"Showdown in Marlboro Country," Consumer Reports,
September, 1969.

"Artificial Ocean Attracts Surfers to Arizona Desert," Big Surf, Tempe, Arizona.

Kenneth Roberts, <u>Leisure</u> (New York: Humanities Press, 1970.)

Stanley Parker, The Future of Work and Leisure (New York: Praeger Publishers, 1971).

Robert Theobald, "The Spectre of a Leisure-Ridden World," The Futurist, June, 1967, p. 36.

Leonard Sloane, "Labor Gives Way to Leisure," New York Times, January 11, 1973.

Frank Trippett (Senior Editor), "The Ordeal of Fun," Look, July 29, 1969.

Marion Clawson and Jack L. Knetsch, "Outdoor Recreation, Resources for the Future Study Poresees Huge Increase," The Futurist, June, 1968, pp. 56-57.

Isaac Asimov, "What will Society be Like in 2068 " Textile World, 118:4:316-321, April, 1968.

<sup>64</sup>Lawrence Lerner, "Computers and Automation," <u>The</u> Futurist, May, 1969.

Richard Kostelanetz, "Metamorphosis of the Arts," The Futurist, February, 1970, pp. 23-24.

Robert Angus and Norman Eisenberg, "Are Cassettes Here to Stay?" High Fidelity, July, 1969.

Dennis Gabor, "3-D TV: the Outlook" Popular Science, July, 1966.

Roy Perrot, "Danes Learn to Live with Pornography,"
Los Angeles Times, September 5, 1971.

"A New Word is Born: 'Gyna-cosmetic'," Washington Post, March 8, 1970.

"Advertising: Media Wary of Contraceptive," New York Times, March 11, 1970. <sup>65</sup>John R. Pierce, "Foreseeable Advances in Communication," <u>The Futurist</u>, October, 1967, pp. 80-81.

J.L. Hult, "Cheap Communication," <u>The Futurist</u>, June, 1969, pp. 70-72.

Ithiel de Sola Pool, "Social Trends," Science & Technology, April, 1968.

Peter C. Goldmark, "Tomorrow We will Communicate to our Jobs," The Futurist, April, 1972, pp. 55-58.

"Satellite Communications Raise Hope for Third World," The Futurist, April, 1972, p. 59.

"Information Technology and Its Implications," The Futurist, December, 1972, pp. 244-247.

66 Ben Bagdikian, "The Information Machine," The Futurist, October, 1971, p. 183.

Barry Schwartz (ed.), Human Connection and the New Media (New York: Prentice-Hall, 1972).

Leo Bogart, "The Newspaper of the Future," <u>Seminar</u>, September, 1968.

Morris Janowitz, The Community Press in an Urban Setting (Chicago: University of Chicago Press, 1967).

"Ex libris," Time, February 23, 1970.

"Dealers on Book Row Fear Rent Rises will End an Era," New York Times, September 30, 1969.

67 Louis A. Zurcher, "The Mutable Self," The Futurist, October, 1972, pp. 181-185. Joel C. Snell, "The Urban Student Alienation and School Participation," <u>Urban Education</u>, July-October, 1971, pp. 267-272.

Xerox Corporation, "Life May Become too Easy," The Futurist, April, 1969, pp. 35-36.

Robert F. Kennedy, <u>To Seek a Newer World</u> (New York: Bantam Books, 1968).

Mary Lystad, "Social Alienation: Review of Current Literature," Sociological Quarterly, 13:90-113, Winter, 1972.

Maurice R. Stein, Eclipse of Community (New York: Harper & Row, 1960).

Anton C. Zigderveld, The Abstract Society (New York: Doubleday, 1970).

Robert N. Butler, "The Burnt Out and the Bored," The Futurist, June, 1970, pp. 82-84.

Philip Harris, "From Evolution to Emergence," The Futurist, October, 1967, pp. 74-75.

John Gardner, "How 20th Century Man let His Institutions go to Pieces," <u>The Futurist</u>, December, 1968, pp. 126-128.

Rees McGee, Social Disorganization in America (San Francisco: Chandler Publishing, 1962).

Earl Hubbard, "Mankind in the Universe, an Image of Man for the 21st Century," <u>The Futurist</u>, December, 1967, pp. 90-92.

Robert Z. Apostol, "The Impact of Today's Alienation on the Year 2020," (Private interview). Big Corporations Adopt Psychiatry," New York Times, December 26, 1968.

Ted Landsman, "The Beautiful Person," The Futurist, April, 1969, pp. 41-42.

Willard B. Frick, "The Personal Crises of College Students: Adjustment or Growth?" The Futurist, April, 1968, p. 36.

Stanley F. Yolles, "The Promise of Mental Health," The Futurist, June, 1969, pp. 64-65.

Albert Rosenfeld, <u>II Genesis -- The Coming Control of</u> of Life (New York: Prentice-Hall, 1970).

Jose M. R. Delgado, M.D., "Physical Control of the Mind Toward a Psycho Civilized Society," World Perspectives Series, 41 (New York: Harper & Row, 1970).

Perry London, <u>Behavior Control</u> (New York: Harper & Row, 1970).

The Futurist, April, 1971, pp. 58-60.

"Group Therapy Catches Hold," for more information, National Training Laboratories, National Education Association, 1201 Sixteenth Street, N.W., Washington, D.C. 20036.

71As an overview to the evolution and characteristics of super-industrialism, please see the illustration directly below. In capsule form, this chart outlines the various social and economic features which come to play in past, present, and future societies.

## SOCIETAL TYPE

Social Characteristic	Past Societies s	Present Societies	Future Societies
Dominant Social Relationship	Fellowship Kinship Neighboring Family Law	Exchange Rational Calculation State	Fluid Assoc.
Central Institution	Extended kin group	Social Market or Capitalistic Economy	Social Market or Keynsian Capitalist Economy
Individual & Social Order	Self	Person	Situational Personality
Characteristic Form of Wealth	Land	Money	Experience "New Money"
Type of Law	Family	Contracts	Regional Contract
Order of Institutions	Family Rural Village Town	City Rational Cosmopolitan Life	Serial Family Ad-hocracy Decentrali- zation
Types of Social Control	Concord Folkways Religion	Convention Legislature Public Opinion	Technology Decentralized Democracy

All three of these systems are based on the content of the means of production.

Toffler's system is predicated on affluence, industrialism, urbanism, and rapid change.

John McHale, "Planning the Future," <u>Current</u>, September, 1969, pp. 16-25.

73Lewis A. Coser, Masters of Sociological Thought:

Ideas in Historical and Social Context (Harcourt, Brace,
Jovanovich, Inc., 1971).

Nicholas S. Timasheff, Sociological Theory: Its Nature and Growth (New York: Random House, 1967).

Margaret Wilson Vine, An Introduction to Sociological Theory (New York: David McKay Co., 1967).

Lewis A. Coser and Bernard Rosenberg (eds.), <u>Socio-logical Theory: A Book of Readings</u> (New York: MacMillan Co., 1965).

74 Brian J.L. Berry and Frank Horton, Geographic Perspectives on Urban Systems (Englewood Cliffs: Prentice-Hall, 1970).

John Friedman and John Miller, "The Urban Field,"

Journal of the American Institute of Planners, November,

1965, pp. 312-319.

76 Ibid.

77 Ibid.

78 Ibid.

79 Berry and Horton, op.cit.

80 Carl H. Madden, "Business and the Future," The Futurist, December, 1969.

"Boswash, Chipitts, and Sansan," The Futurist, February, 1968.

Edward E. Booher, "Decades Ahead for a Publisher's View," Science, November 17, 1967.

81 Berry and Horton, op.cit., p. 87.

Phillip H. Rees, <u>Factoral Ecology of Metropolitan</u>
<u>Chicago</u> (Chicago: University of Chicago, 1968).

83 Ibid.

84E.T. Hall, <u>The Silent Language</u> (New York: Doubleday, 1959).

Brown & Co., 1970).

Banfield, The Nature and the Future of our Urban Crises, the Unheavenly City (Boston: Little, Brown & Co., 1970).

86 Ibid.

87 Rees, op. cit.

88 Ibid.

89Wondrous World of 1990: Outlook for Young People,"
U.S. News & World Report, January 30, 1967, pp. 62-67.

"Changes in State Populations by Race," Statistical Bulletin, Published by the Metropolitan Life Insurance Company, New York, May, 1969.

90 Anthony Downs, " Alternative Futures for the American, " Daedalus, 97:4:1331-1378, Fall, 1968.

91 Rees, op. cit.

Chapter Three

Population and Human Ecology

## Population and Human Ecology

# Introduction

population and ecological considerations have been the concern not only of sociologists, demographers, and human biologist, but an emerging number of geographers. Our especial concern in this chapter will include the impact of the environment on urban form.

Emerging years ago, but:only recently noted by the general public, two opposing camps developed concepts regarding the question of population increases. The Neo-Malthusians maintained that not only was the world experiencing major increases in population, but that the United States also had a population problem.

They also maintained that population acts upon the system as a significant intervening variable. They indicated that each new increase of people, given present consumption rates, made tremendous demands upon the environment. For example, they contended that as population increased by one per cent per year, energy usage increased about seven per cent. Solid wastes per person have

doubled since the 1920's and the population has tripled since 1900, with the country now using more than 50 per cent of the world's resources.

Others contest this position essentially upon the basis that the man per acreage ratio in the United States, relative to western Europe, is quite low, and that the United States is prosperous enough to support a considerable increase in population. 2

In terms of the current population, the "Right to Life" people note the falling rates of population increase (down considerably from the beginning of American history with the exception of World War II) and the "ZPG" people note the real number increases (an additional 2-3 million people each year).

Notwithstanding the ideological aspects of population changes, how does this translate into growth for the world and this country?

Miles<sup>3</sup>, as he views the world population sees what he feels are alternatives most likely to occur. These he labels a "modified Irish curve." In this he sees a number

of severe disasters but not of calamitous nature, which would would result in a downward population curve so that by 2020 world population would revert to the 3-4 billion level we have today.

Questions may be raised about the growth of this country. The present population is approximately 210 million people. The Population Reference Bureau no longer projects one population figure, but a series of figures based on varying assumptions from 440,253,000 as a high to a low of 270,447,000 by 2020. Even with a very conservative estimate of population it would appear that there will be an additional 60 million people living in the United States in 2020.

The superindustrial state is a very mobile one, and projections indicate that migration also will increase.

Mortality rates may also be decreased, thus causing the aged to become a more viable political and social entity.

## Population Distribution

Given these trends, the population commission projects that by the year 2000, 85 per cent of the population will live in metropolitan regions. By the turn of the century,

the metropolitan regions will have an increase of 40,000,000 persons. (Assuming a 2-child family.) They see the super-industrial state forming into megapolitan regions, with workers willing to commute much further geographically because of high speed transit systems. They envision:

Urban regions appear to be a prominent feature of the demographic future, and if our population distributes itself according to projections, 54% of all Americans will be living in the two largest urban regions — one stretching along the Atlantic seaboard and westward past Chicago containing 41% of our population, the other in California between San Francisco and San Diego, containing 13 per cent. 5

## Resources and Environment

Notwithstanding the ideology of population change, there appears to be a decided relationship between resources and population. As the Commission on Population notes:

Population growth is one of the major factors affecting the demand for resources and the deterioration of
the environment. The further we look into the future,
the more important population becomes.

Some sources see the environment deteriorating on a world scale, and with this a growing feeling of national geographic interdependence. 7 Importantly one of the inadvertent problems is the effect upon the world's climate.8 Ecologists contend that, in the past, producers have not taken into account the extenualities or the cost of maintaining the environment. They see the effects of water pollution with infiltration of raw sewage, oil, and harsh . pesticides into inland streams. They also indicate that air is heavily infiltrated by particulates and lead. Of some 200 million tons of waste poured into the air last year, the auto contributed 94.6 million tons. They also see increases in noise pollution, and due to inadequate recycling and sanitation land fills, more solid waste problems.9

However, not all concur. Some see the environmental issue as already on the way to a solution due to aggressive action or because the issue was not significantly important in the first place. Others see emphasis on environmental concerns causing unemployment and a slowing of

industrial activity. Still others see the environmental movement as a cover issue for the promoters of a left-wing government. Denergy is also involved in this issue. Cook and Waring note that throughout history the industrial age has been the most devastating in terms of consumption of resources. 11

Others feel that major sources of pollution of air, water, and soils can be directly or indirectly related to the nation's energy demands and the economics of energy. They contend that much of this energy use could be substantially reduced through effecting some changes in economic and social patterns on a national as well as a local basis.

Water pollution and energy efficiency relationship, they would contend, can be illustrated in many ways.

The power generating plant using water as a condenser coolant introduces thermal pollution. The greater the demand for electrical energy, the greater the output of thermal pollutant. The manufacturing plant that uses water in washing or in chemical processing can frequently

relate its product and its process to energy use . . .

the principle of built-in obsolescence requires the

frequent replacement of a product at a high energy cost

. . . the generally considered low cost of water and its

seemingly unlimited supply provides an economic climate

for abuse.

They would indicate that much can be said for increasing the sources for energy to meet the growing demands made by our society, but regardless of all efforts in inputs, more intensive exploration, trans-Alaska pipelines, accellerating strip mining, gasification of coal, etc. the fact remains that this earth is finite and sooner or later we will have used up the last drop of oil, the last hod of coal. Thus, a long-range effort at conserving fuels is urgently needed along with technologies that enable us to utilize renewable energy resources: water power, tidal power, wind, and possibly geothermal energy. 12

However, not all environmentalists concur. Some feel that the environment "crisis" is overstated or will be effectively circumvented by technology. For future energy

they see the viability of new energy sources with especial emphasis on nuclear reactors. Tapping solar energy is also thought possible as is the renewed use of coal. Though present projections for certain energy sources show diminished abailability, other sources can quickly take their place. 13

Environmentalists also see problems with nuclear reactors in terms of shipment, air pollution, potential
nuclear disaster, and emissions of plutonium oxides. 14

## Food Supply

Though Ehrlich 15 has raised the issue that a doubling of population by the year 2000 (to 7 million) will outstrip the food supply, Brown 16 is more optimistic. Since 1967, Brown notes a new wheat strain, I-38, has been developed and proved successful in Asia. Some look to aquaculture as a source of world food supply, and still others see productivity of artificial meats. 18

The impact caused by changes in the supply of food available to this country is more political than ecological.

Though the United States is quite affluent, people in some 280 counties of the nation's 3,100 counties suffer from hunger and malnutrition. 

19 However, the super-industrial state may require even more food as American consumption "needs" increase. If projections that poverty will be eradicated are true, hunger should be considerably diminished.

### Human Biology and Urban Form

How do these factors concerning the physical and ecological environment translate into urban form?

One source notes that cities may best be viewed as "condensed regions of population." Population distribution has been described for the American city as a distributed sheet. As Brown notes:

Density: Despite more people in metropolitan areas, there has been a lesser concentration of population density within these areas. This is because of expanded utilization of the land areas in suburban rings by an increase in the number of people living in these outlying areas. This flight to suburbia has been made possible through

advances in technology. One of our studies shows that only about 1/5 of all commuters spend more than a half hour getting to work. 21

It would appear as population density increases to threshold size (15,000 to 35,000 dependent upon cartographic and physiographic impediments), horizontal expansion will occur. A "quasicolloidal" dispersion occurs in an amoebalike fashion radiating from the core of the city. In the future, given super-industrial trends, a further scatteration of population with heavy losses in core city and rural areas will occur in spite of the effect of natural increases.

Population distribution has been measured by densitydistribution curves which measures human numbers as they
are distributed from core to rim. These curves indicate
selected impaction in low income areas and other critical
density regions.

One of the key features of population distribution is the migratory processes that shape new man-to-land formations. Inter-urban moves are most likely to come from young white collar and manual workers, as well as unemployed youth.

Evidence indicates that moves are likely to occur if
there is a life cycle change in the family, improvement or
diminishment of income, ethnic or racial change in the
neighborhood, or a necessary personal adjustment such as
conflict with neighbors. Though the data is descriptive,
it appears that when people move, they are likely to choose
a home that is propinquitous to their previous home. They
also consider access to the rest of the city and the distance of their work journey.

As an overview, it appears that distribution in the American city system follows both social and economic considerations discussed earlier. On balance, there is continual out-flow of numbers of people from core to rim as the city ages. Increases in human numbers lead to increases in land consumption.

What would happen if there were considerable effort to reduce population, preserve the environment, and encourage a system based on "social profit"?

## ZPG Society

John R. Meyer, a Yale economics professor and president of the National Bureau of Economic Research, says that predicting the future under the best circumstances is difficult for economists. 22 Nevertheless, he ventured this scenario: take-home pay in a ZPG society should improve because of reduced taxation. Savings should increase.

Early retirement should become more common.

The Commission on Population Growth on which Meyer served predicted that regardless of the birth rate, average family income will increase from today's \$12,000 to more than \$21,000 (in 1972 dollars) by the year 2000, even assuming that the work week drops to 30 hours. 23

The fertility rate decreased from 3.77 children

per family in 1957 to 2.03 in 1972, below the replace
ment level of 2.1. If that low rate persists, one

result will be a gradual balancing of age groups

within the populace... The median age will gradu
ally increase. It is now 28, but at a 2.1 fertility

rate it would reach 34 by the year 2000. Economists

say that this will mean more workers with fewer

dependent mouths to feed, and hence greater affluence.

The U.S. Commission on Population Growth and the Ameri
can Puture estimated that per capita income in the

year 2000 may be 15% higher with a birth rate of 2 children per family than it would if births averaged 3 per family. 24

J.J. Spengler of Duke University writing in the Population Reference Bulletin discusses both the advantages and disadvantages of a Zero Population Growth Society. 25 Listed among advantages, are a more affluent system caused by diminished age-dependency ratios, and more stable consumer demand from more regulated trade-cycle dependencies. He also raises the possibility of more family stability due to a reduction in the present 13 to 18% of unwanted births. The system would probably result in improved manenvironment relationships bringing about increased improvement in time which would be used seeking other goals than growth. Importantly for the structure of urban form, he states:

Since population concentration, often accompanied by mal-distribution in space, is a concomitant of population growth, the slowing down of this growth will make possible improvement in the spatial distribution of the population. Interstate differences in population density, when corrected for differences in resource structure (e.g., water supply), indicate our population to be quite mal-distributed and destined to become more so should the anticipated increment -- 75-100 million inhabitants and 30-40 million jobs -- be distributed at all like our present 208 million.

While this increment will settle in cities and nearby rural areas, the location of these cities need not be dominated by random processes and the continuation of our goalless system of national transportation. Man no longer is bound to the soil and mineral deposits. Moreover, while he is not free to locate at will, the options available to him are many and can be increased by miniaturizing plants of optimum size. Most men, being "job-takers," must, however, settle where "job-makers," provide employment.

Since about one job in four is provided by 750-850 business firms managed by perhaps 10,000 "strategic

decision-makers." Collective locational freedom is very great, particularly given supplementary assistance at the hands of public corporations and private enterprisers, mainly in the form of infra-structure and residential facilities. These strategic decisionmakers should be able, over the next 50 or more years to establish self-sustaining bases for 400 to 500 or more new cities, in less populated states, away from currently growing concentrations destined to be short of water as well as space. The new cities could accomodate both the prospective increase in the nation's urban population and the increasing number who are fleeing the disadvantages, including high vulnerability to the thermo-nuclear extinction, associated with today's metropolitan and megalopolitan concentrations, many of which are collapsing under the weight of economically dysgenic social metabolism.

However, he also sees problems, First, an aged population (some 11-17 % of the population) increases the demand for social welfare. Second, advancement within formal associations will be more difficult. In terms of the labor force,

slowing down the rate of population growth may intensify the task of maintaining inter-industrial, inter-occupational, and inter-regional balance. He also envisions political struggle concerning income distribution in nogrowth system, and controversy over higher inflation.

Locally, a reduced population would free income spent on costs necessary to provide quality education. In terms of construction of dwelling units, the city would have reduced start-up costs for sewers, streets, and other city services.

The city would probably have less expansion. Welfare costs would be diminished. And perhaps, the form itself would emerge as Spengler describes, but scatteration
and population growth are not necessarily synonymous.

Energy shortages could encourage large portions of population to revert back to less energy consuming mass transit. Energy concerns would be a smaller problem as a result of diminished population.

### Pight to Life

The development of this system is harder to create. Wilkes 26, has pointed out birth control is not an issue for those people who do not see societal problems in a

highly populated country. Though opposed to abortion, wilkes does not support (but does not oppose) family planning. Given the falling birth rates which have persisted since the establishment of this country, a high population rate increase is less-likely.

#### As Samuelson notes:

Certainly improvements in techniques of birth control play a role. The Pill, loop, sterilization, legal abortions, all contribute to the decline in births. A presumed increase in frequence of sexual intercourse is more than offset by effective use of birth control.

The loosening up of the Catholic Church since

Pope John has meant both a sharp drop in aspiration

goals of Catholic couples and a sharp rise in their

ability to achieve their projected goals for family

size. Another generation of convergence between

Catholics and the rest of the population in these

matters would result in practically no remaining

differentials.

highly populated country. Though opposed to abortion, wilkes does not support (but does not oppose) family planning. Given the falling birth rates which have persisted since the establishment of this country, a high population rate increase is less-likely.

#### As Samuelson notes:

Certainly improvements in techniques of birth control play a role. The Pill, loop, sterilization, legal abortions, all contribute to the decline in births. A presumed increase in frequence of sexual intercourse is more than offset by effective use of birth control.

The loosening up of the Catholic Church since

Pope John has meant both a sharp drop in aspiration of

goals of Catholic couples and a sharp rise in their

ability to achieve their projected goals for family

size. Another generation of convergence between

Catholics and the rest of the population in these

matters would result in practically no remaining

differentials.

Birth rates of Blacks, although still higher than those for Whites, are also converging down. The differentials that existed in the 1920's between the rural masses and the educated middle classes are also narrowing significantly. 27

Therefore, the likelihood of a high populated and impacted situation similar to that found today in western Europe is less likely but is still possible.

If there is a return to the traditional role for women, a value orientation toward large families, and the attraction of high turnover industries requiring manageable pools of labor (as has occurred recently in Japan), fertility rates may increase.

What type city would this bring about? The sprawl may scatter to the limits of the city. Start up costs would mount, tax per person per thousand would be much higher.

Land developers would prosper and the age-dependency ratio would increase. There would be a booming youth market, but there would also be a selected interim energy shortage crisis, and unless life styles and consumption patterns changed radically, more pollution.

In-summation, however, it would appear that population will stabilize but the real number growth will still place demands upon the environment. Energy will probably not be a crisis except upon an interim basis because of America's economic position and military dominance. Notwithstanding ideology, it is assumed by these authors that large dominant nations will not suffer to the degree other nations will. Secondly, the nation will be using nuclear reactors. Though plutonium has a finites life, the reactors will be erected in spite of litigation presented by groups such as Ralph Nader and the Friends of the Earth. The reactor will probably produce the energy needed but will appear to place much of the population in real jeopardy. However, we see AEC as having the political muscle to carry it out over and above its opposition.

The environment on balance will probably continue to suffer given American life style and socio-economic system. New definitions may come into operation about an industrial or super-industrial environment. As William Wise notes in <a href="Killer Smog28">Killer Smog28</a> new generations can define old problems in new ways even if the problem becomes more severe.

#### Footnotes

Paul R. Ehrlich and Ann H. Ehrlich, Population Re-Sources Environment (San Francisco: W.H. Freeman, 1970).

"Population Growth America's Future" An Interim Report Prepared by the Commission on Population Growth and the American Future, Superintendent of Documents, U.S. Government Printing Office, Washington, D.C. 20402, 1972.

Caroline Bird, "The Crowding Syndrome, Human Rights: Man is Losing Some But Gaining Others," The Futurist, February, 1973, pp. 33-34.

Eugene P. Odum, et al., The Crisis of Survival (Madison: Scott, Foresman, and Company, 1970).

Paul R. Ehrlich, The Population Bomb (New York: Ballantine Books, 1968).

Garrett Hardin, Stalking the Wild Taboo (Los Altos: William Kaufmann, Inc., 1973).

G.H. Dunne, "Development: Christian Concern?" America, 127:466:-469, December 2, 1972.

Albert L. Kraus, "The 1960's Should Have Been No Surprise for the Demographers," New York Times, XII:16-2, January 11, 1970.

Leslie and Charles S. Westoff, From Now to Zero (Little, Brown, 1971).

Isaac Asimov, "Can Man Survive?" True Magazine, January, 1971.

J.M. Tanner, "Earlier Maturation in Man," Scientific American January, 1968.

"Zero Population Growth," Los Altos, California, or Planned Parenthood, World Population, New York, N.Y.

"Tax Incentives Proposed to Limit Family Size," Wash-ington Star, February 23, 1970.

"Crowding Out," Washington Post, Pebruary 27, 1970.

"They Chose to be Childless," New York Times, March 23, 1970.

E. James Lieberman, "The Case for Small Families," New York Times, March 8, 1970.

"Population Will Strain Natural Resources," The Futurist, December, 1969, p. 156.

Sue Titus Reid and David L. Lyon, Population Crisis and Interdisciplinary Perspectives (Glenview: Scott, Foresman and Co., 1972).

<sup>2</sup>Ben J. Wattenberg, <u>This U.S.A.</u> (New York: Pocket Books, 1966).

Dr. and Mrs. J.C. Wilke, Handbook on Abortion (Cincinnati: Hiltz Publishing, 1972).

Paul Johnson, "The Next Hundred Years," New Statesman, March 28, 1969, pp. 438-439.

R.M. Scammon, "This is What a Baby can Look Forward to," Life, 63:28-29, December 1,1967.

John Maddox, <u>Doomsday Syndrome</u> (New York: McGraw Hill, 1973).

Wilfred Beckerman, "The Myth and the Environment Catastrophe," National Review, November 24, 1972, pp. 1293-1295.

W.L. Wearly, "Business Tells it Like it is," (Commence-ment Address) Wilkes College, Willkes-Barre, Pa., June 4, 1972.

Gene Sheppard, "The Disaster Lobby on the March," Human Events, March 6, 1971, pp. 1-4.

Gary Allen, "Environment: Doomsday and Dictatorship," American Opinion, February, 1973, pp. 1-18.

"Changes in the Average Number of Children Ever Born to Women, 1960-69," Census Bureau Report, Superintendent of Documents, Government Printing Office, Washington, D.C., 1970.

"France Blames Accidents, Alcoholism, and Insecurity for Underpopulation," New York Times, April 13, 1970.

Rufus E. Miles, "Three Ways to Solve the Population Crisis," Man's Population Predicament -- Population Bulletin 27:2, April, 1972 (Published by Population Reference Bureau).

The Future Population of the United States, Popula-Reference Bureau, Inc., Washington, D.C., 1972.

<sup>5</sup>The Commission on Population Growth and the American Future, Population and the American Future, U.S. Government Printing Office, Washington, D.C., 1972.

6 Ibid.

7"The Planetary Perspective of John McHale," The Futurist, December, 1970, pp. 202-204.

Irwin Hersey, "We will have to Run Very Hard Just to Stay Even," Engineering Opportunities, May, 1969.

Rachel Carson, Silent Spring (Boston: Houghton, Mifflin, 1962).

Paul R. Ehrlich, op.cit.

Gordon Taylor, The Doomsday Book: Can the World Survive? (New York: World Publishing Co., 1971).

Dennis L. Meadows, "Predicament of Mankind," The Futurist, August, 1971, pp. 136-144.

Donnella H. Meadows, et al., The Limits to Growth (New York: Universe Books, 1970).

J.W. Forrester, World Dynamics (Cambridge: Allyn press, 1971).

Charles W. Williams, Jr., "Inventing a Future Civilization," The Futurist, August, 1972, pp. 137-141.

John R. Platt, "Masterminding Evolution: Man's Godlike Task," The Futurist, April, 1968, pp. 30-32.

Barry Commoner, Science and Survival (New York: The Viking Press, 1966).

- B. Fuller, "Report on the Geosocial Revolution," Saturday Review, 50:31-32, September 16, 1967.
- 8"Man's Inadvertent Alteration of the World's Climate,"
  The Futurist, June, 1972, p. 122.
- J. Murray Mitchell, Jr., "Is Man's Industry Upsetting World Weather?" The Futurist, "April, 1969, pp. 34-35.
- Olifford E. Knapp, "Ecology Series," Instructor Publications, 1970.

Leslie L. Roos, Jr., (ed.) The Politics of Ecosuicide (New York: Holt, Rinehart, and Winston, 1971).

Paul Swotek, The Users Guide to the Protection of the Environment (New York: Valentine Book, 1970).

Walt Anderson (ed.), Politics and Environment: A
Reader in Ecological Crisis (Pacific Palisades: Goodyear
Publisher, 1970).

Paul Ehrlich and Richard L. Herremon, How to Be A Survivor, (New York: Valentine Book, 1971).

Rachel Carson, op.cit.

Sue Titus Reid and David L. Lyon, op.cit.

"Needed: Better Regulation of Pesticides and Integrated Methods of Pest Control," Published by the Conservation Foundation, Washington, D.C., 1970.

"Swing to Unleaded Gas: What's Ahead," U.S. News and World Report, December 28, 1970, p. 28.

"Payrolls and Pickerel in Maine," Time, February 16, 1970, p. 52.

Walter J. Hickel, Who Owns America? (New York: Prentice-Hall, 1972).

"Ecology Becomes Burning Issue," See National Parks Association, Washington, D.C.; Sierra Club, San Francisco, California. The Conservation Foundation, Washington, D.C. Envi ronmentalControl Administration, U.S. Department of Health, Education, and Welfare, Rockville, Maryland. The National Wildlife Federation, Washington, D.C.

William Wise, op.cit.

"New Bag on Campus," Newsweek, December 22, 1969, p. 72.

"The Politicians Know an Issue," Newsweek, January 26, 1970, p. 33.

A Primer for People Who Care! What you can do to save your Environment, Nebraskans for Peace (Locally Published).

Theodore H. White, "How do we Get from Here to There?" Life, 1970.

Nature's Living Museum, Mutual of Omaha (Locally Published).

"An Environmental Bibliography," Editorial Research Section, Office of Public Affairs of the Environmental Protection Agency, Washington, D.C. 1971. (Please see this for a listing of over 25 different sources).

"Business Gets Into Social Indicators," Behavior Today, March 6, 1972

"Troubled Little Island," Time, January 26, 1970, p. 55.

Barry Commoner, Science and Survival (New York: Viking Press, 1969).

"Heat Pollution -- or Enrichment," <u>Industrial Research</u>, July, 1968.

John Clark, "Thermal Pollution: A Threat to Cayuga's Waters?" Science, 162:649-50.

John Clark, "Thermal Pollution and Aquatic Life," Scientific American, March, 1969.

"Wintertime Mini-Crisis," Newsweek, February 15, 1971, pp. 73-74.

"Heat, Smog, Power Crisis: Where, and Why," U.S. News & World Report, August 10, 1970, pp. 14-16.

Fair Stand Committee, Federal Water Quality Advice (Locally Published).

"Conservation Caretaker," Time, February 9, 1970, p. 46.

"Here Comes the Smog," Newsweek, August 10, 1970, pp. 64-65.

"Nixon Starts the Cleanup," Time, February 23, 1970, pp. 39-40.

Robert H. Anderson and Larry C. Holcomb, "Quality Environment Council," Pamphlet (Locally Published).

"The Advent of Big Biology," Time, March 23, 1970, p. 40.

10 John Maddox, op.cit.

Wilfred Beckerman, op.cit.

W.L. Wearly, "Business Tells it Like it is," op.cit.

Gene Sheppard, op.cit.

Gary Allen, op.cit.

Victor Ferkiss, <u>Technological Man:</u> The Myth and the Reality, (George Vraziller Publisher, 1970).

"The Return of the Returnables?" Newsweek, September 21, 1970, pp. 70-71.

"U.S. Industry Makes Amends," Newsweek, January 26, 1970, pp. 45-47.

"Pollution: Growing Menace -- What U.S. Is Doing About It," U.S. News & World Report, June 9, 1969, pp. 40-43.

"Nixon's Spirit of '76," Newsweek, February 2, 1970, pp. 17-18.

"Latest Moves on Pollution Control," U.S. News & World Report, February 2, 1970, p. 5.

"The People's Protector," Time, March 16, 1970, p. 62.

"Ford's Better Idea," Time, December 19, 1969, p. 51.

"The Rise of Anti-Ecology," Time, August 3, 1970, p. 42.

Aaron Teller, "The Only Way Out of Pollution," The American World Report, August 31, 1970, pp. 43-44.

"Now a Boom in Devices to Fight Pollution," U.S. News & World Report, August 31, 1970, pp. 43-44.

Ralph Hamil, "The Future of the Oceana," The Futurist, June, 1972, pp. 109-110.

"Endorsing Infanticide?" Time, May 28, 1973, p. 104.

"How to Stop Pollution," U.S. News & World Report, November 23, 1970, pp. 54-57.

11Earl Cook, "Energy Sources for the Future," The Futurist, August, 1972, pp. 142-152.

John Waring, "The Horsepower Explosion," The Futurist, February, 1969, pp. 23-24.

12Professor George Grube, Head of the Environmental Sciences Area of the Biology Department at Dana College, and a member of the Quality Environmental Council.

Philip H. Abelson, "Energy Conservation," Science, 1972, p. 355.

Philip H. Abelson, "Energy and National Security," Science, 179:857, 1973.

Duan Chapman, Timothy Tyrrell, and Timothy Mount, "Electricity Demand Growth and the Energy Crisis," <a href="Science">Science</a>, 178: 703-708, 1972.

John T. Edsall, "Nuclear Energy," Science, 178:933, 1972.

Robert Gillette, "Nuclear Safety: Damaged Fuel Ignites a New Debate in AEC," Science, 177:330-331, 1972.

Robert Gillette, "Nuclear Safety (I): The Roots of Dissent," Science, 177:867+871, 1972.

Robert Gillette, Nuclear Safety (II): The Years of Delay, "Science, 177:771-776, 1972.

Robert Gillette, "Nuclear Safety (IV): Barriers to Communication," Science, 177:1080-1082, 1972.

Robert Gillette, "Reactor Safety: AEC Concedes Some Points to Its Critics," Science, 178:482-484, 1972.

Robert Gillette, "The Budget of the U. S. Government, 1974: Energy," Science, 179:549-550, 1973.

Allen L. Hammond, "Energy Options: Challenge for the Future,: Science, 177:875-876, 1972.

Allen L. Hammond, "Solar Energy: The Largest Resource," Science, 177:1088-1090, 1972.

Toward Balanced Growth: Quantity with Quality, Report of the National Goals Research Staff, (Washington: U. S. Government Printing Office) July 4, 1970.

"The Energy Crisis: Time for Action," Time, May 7, 1973, p. 41.

"Electric and Solar Power Consumption May Rise," The Futurist, August, 1967).

"It Can't Be Ignored," Midwest Focus on the News, Omaha World Herald, Omaha, Nebraska, March 1, 1972, p. 37.

"Goals and Objectives Suggested by Citizens at an Open Meeting of the Quality Environment Council and the Soil and Water Conservation Service," January 13, 1973.

Allen L. Hammond, "Fission: The Pro's and Con's of Nuclear Power," Science, 178:147-149, 1972.

Allen L. Hammond, "Conservation of Energy: The Potential for More Efficient Use," Science, 178:1186-1188, 1972.

- Allen L. Hammond, "Energy Needs: Projected Demands and How to Reduce Them," Science, 178:1186-1188, 1972.
- Allen L. Hammond, "Energy and the Future: Research priorities and National Policy," Science, 179:1116, 1973.
- Allen L. Hammond, "Solar Energy: Proposal for a Major Research Program," <u>Science</u>, 179:1116, 1973.
- Use in the United States, "Science, 179:1299-1304, 1973.
  - Constance Holden, "New Energy Message Downplays R & D," Science, 180:475, 1973.
  - G. A. Lincoln, "Energy Conservation," Science, 180: 155-162, 1973.
  - Thomas H. Maugh, II, "Fuel from Wastes: A Minor Energy Source," Science, 178:599-602, 1972.
  - Dennis L. Meadows, "Nuclear Energy and Growth," Science: 179:855-856, 1973.
    - Stephen P. Potter, "Energy Supply," Science, 180:446, 1973.
  - 13 David Brand, "Catching Sunbeams -- Old Dream of Putting Sun's Fire to Work Gets Renewed Attention," Wall Street Journal, April 16, 1973, pp. 1 and 14.
  - "'Pig Power' as Motor Fuel is High-Octane, Nonpolluting," Omaha World Herald, April 22, 1973, p. 4A.
  - Glenn T. Seabord and William R. Corlis, New York: E.P. Dutton & Co. Man and Atom: Building a New World Through Nuclear Technology, 1972.
  - Glenn T. Seaborg, "Some Long-Range Implications of Nuclear Energy," The Futurist, February, 1968, pp. 12-13.

"Wondrous World of 1990: Outlook for Young People,"
U.S. News & World Report, January 30, 1967, pp. 62-66.

Harrison Brown, James Bonner, and John Weir, The Next Hundred Years (New York: Viking Press, 1966).

Glenn T. Seaborg, "The Birthpangs of a New World," The Futurist, December, 1970, pp. 205-208.

Lt. Col. Joseph Martino, "Forecasting Technological Break-throughs," The Futurist, June, 1970. pp. 101-102.

"Energy and the Environment," (speech) MBAA meeting in Chicago, May, 1970.

"Nuclear Power 1973-2000," (Report) Atomic Energy Commission, Forecasting Branch, Office of Planning and Analysis.

Turner, "The Long Future of the Economy," op.cit.

David A. Burkwitz and Arthur M. Squires (eds.) Power of Generation and Environmental Change (Cambridge: MIT Press, 1971).

"Utah International Mines Coal, Iron, Copper, Uranium," <u>Investor's Reader</u>, April 4, 1973, pp. 16-19.

Gerald Feinberg, "Long-Range Goals in the Environment," The Futurist, December, 1971.

14 "Opposition to Breeder-Reactor Mounts," The Environmental, Vol. IV (Columbus, Ohio: Earthday Committee of the Ohio State University, October, 1972).

"Special Issue: Ecopornography," Rodale's Environmental Action Bulletin, Vol. III (Emmaus, Pa.: Rodale Press, Inc.).

David Brower, John Holdren, Mr. Lovins, and Jerry Weingart, "How to Use Less Energy," Not Man Apart, Vol. II (San Francisco: Friends of the Earth, August, 1972).

David Brower, "Should Atomic Energy Go Back to the Drawing Board?" Not Man Apart, Vol. II (San Francisco: Friends of the Earth, August, 1972).

Gene Bryerton, Nuclear Dilemma (New York: Ballantine Books, 1970).

Wilson Clark, "Nixon on Energy," Not Man Apart, Vol. II (San Francisco: Friends of the Earth, August, 1972).

Wilson Clark, "Nuclear Plants are Unsafe," Not Man Apart, Vol. I (San Francisco: Friends of the Earth, July, 1971).

Gene G. Downs and Charles P. Norman, <u>Nuclear Power</u>
Controversy, (Farmingham: Eastern Mass. Regional Library
System, 1971).

Robert Gannon, "Atomic Power--What are the Dangers?"
Science Digest, Vol. LXXI (New York: The Hearst Corporation, January, 1972).

Donald Geesaman, "And Now for a Little Diversion...,"
Environment, Vol. 14, (St. Louis: Scientists' Institute
for Public Information, October, 1972).

John W. Gofman and Arthur R. Tamplin, Poisoned Power: The Case Against Nuclear Power Plants (Emmaus, Pa.: Rodale Press, Inc., 1971).

D. D. Grimmer and K. Luszczynski, "Lost Power," Environment, Vol. 14 (St. Louis: Scientists' Committee for Public Information, April, 1972).

David B. Large, "Conserving Energy: Some Things that can be Done," Not Man Apart, Vol. III (San Francisco: Friends of the Earth, March, 1973).

Hugh Nash, "Message from the Future," Not Man Apart, Vol. III (San Francisco: Friends of the Earth, February, 1973).

Jim Rathlesberger, "Administration Watching -- Nuclear Power," Not Man Apart, Vol. I (San Francisco: Friends of the Earth, October, 1971).

The Rand Corporation, "Rand Warns: Technology Offers no Short- or Medium-Term Solutions to the 'Energy Crisis,' So Demand for Power must be Curtailed," Not Man Apart, Vol. III (San Francisco: Friends of the Earth, February, 1973).

The Union of Concerned Scientists, "An Evaluation of Nuclear Reactor Safety," Not Man Apart, Vol. II (San Francisco: Friends of the Earth, August, 1972).

15 Paul Ehrlich, The Population Bomb (New York: Ballantine Books, 1968).

16Lester R. Brown, "The Optimistic Outlook for World Food Production," The Futurist, August, 1969, pp. 89-92.

Lester R. Brown, "New Grain Seeds are Changing the Outlook for Feeding the World's Spiraling Population," Foreign Affairs, July, 1968.

17 Dr. Clarence P. Idyll, Chairman, Division of Fishery Science, Institute of Marine Science, University of Miami, Miami, Florida.

Lloyd B. Stover, "Ocean Objectives: A Porecast to the Year 2000," The Futurist, June, 1970, pp. 89-92.

"Villages on Ocean Floor are Foreseen," The Futurist, June, 1970, p. 93.

"To New Atlantis by 'Submovile'," The Futurist, January/February, 1967, pp. 3-4, 29-30.

Edmund A. Gullion (ed.), Uses of the Seas, (Engle-wood Cliffs: Prentice-Hall, 1968).

Charles B. Jackson, "Tektite Unlocks Secrets from the Ocean's Floor," National Wildlife, 9:3:51-54, April-May, 1971.

Marine Resources and Legal-Political Arrangements for their Development, Commission on Marine Science, Engineering and Resources Panel Report, Vol. 3 (Washington: U.S. Government Printing Office, 1969).

"New Products -- Fish Protein," Sweden News, 6,6,14,

Roger Nevelle and others, The Ocean (San Francisco, W.H. Freeman and Co., 1969).

Eric Sharp, "Seafood Scientists Hoping Experiments Bloom Into Industry," Omaha World-Herald, February 25, 1973.

Peter Tonge, "Ocean Plenty: Fact or Pantasy," The Christian Science Monitor, Midwestern Edition, 65:79:5, February 28, 1973.

Peter Tonge, "Need Chickens Eat Better than People," The Christian Science Monitor, Midwestern Edition, 65:80: 7. March 1, 1973.

Peter Tonge, "Briny Furrows Yield Bumper Crops,"
The Christian Science Monitor, Midwestern Edition, 65:81:9,
March 2, 1973.

Peter Tonge, "The Abundance of the Oceans is NOT Limited," The Christian Science Monitor, Midwestern Edition, 65:82:9, March 3, 1973.

John G. Vanderwalker, "Sciences Window on the Sea," National Geographic, 140:2:256-289, August, 1971.

Lionel A. Walford, Living Resources of the Sea (New York: The Ronald Press Co., 1958).

For further information, see Tatsuo Tsukamoto, manager, Protein and Foods Division, Misshin Flour Milling Company, l Tokyo, Japan.

18"A Nation within a Nation," Time, May 17, 1968, pp. 24-32.

"Challenging the Pharach," Time, May 17, 1968, p. 35.

"Poverty: Who are the Hungry?" Newsweek, April 28, 1969, pp. 38-39.

19 Geographic Perspective on an Urban System with Integrated Readings (Prentice-Hall, 1970).

- 20 "A ZPG Economy," Money Magazine, May, 1973.
- 21 Ibid.
- 22Ibid.
- 23 Ibid.
- 24 Ibid.
- 25J.J.Spengler, "Economic Growth in a Stationary Population," <u>Population Reference Bureau</u>, July, 1971.
- 26Dr. and Mrs/J.C. Wilkes, Handbook on Abortion
  (Cincinnati: Hiltz Publishing Co., 1972).
- <sup>27</sup>Paul A. Samuelson, "Falling Birth Rates," <u>Newsweek</u>, June 19, 1972, p. 75.
- 28William A. Wise, <u>Killer Smog</u> (New York: Audubon Ballantine Book, 1970).

Chapter Four

Geo-Politics

### Geo-Politics

### Political Culture

Geo-politics is used here in its broadest sense to connote how the political structures emerge and interact with man-land relations. The word was originally coined by MacKander but its narrow definition does not apply here.

The political culture is a reflection of an early beginning of projected propertied revolutionaries creating an anti-monarchial private enterprise republic. The ideological base was made with two political parties emerging with striking continuities in platform. Not being exposed to other ideologies such as France or Italy were in their history, the two were non-doctrinaire parties and ensuing governmental policies increased America's position from a small isolationist agrarian republic to a large industrial system with a vast military and political dominance. 2

In viewing the political culture not all agree about its functions. The political Right see a market system that has become protectionist for the large corporation, driving out the small entrepreneur. They feel that the only system that is viable is a governmental operation which remains in a low profile. They believe that the market system will then force decentralization.

On the other hand, the Left sees a system increasingly centralized and as the natural function of the market system. Importantly, they see the government as an arm of the economic structure, i.e. owners of large sectors of productive property. Notwithstanding America's mean wealth, they maintain distribution does not cover various sectors of the economy. 4

Both the Right and the Left share the vision that a small elite dominate the system. They feel that many agencies and institutions prevail to protect a small number at the top. Both maintain that the democratic processes of the system are corrupted by the existing party system and the resulting political network. Both feel that the two main parties are the same or similar. The Right sees the parties leading the country toward socialism and later communism. The Left sees the parties leading the

system toward a fascist state but with outward forms of liberal rhetoric as a cover. Both feel that the elite tyrannize and mislead the political mainstream. To both the center is dead center, a force from resistance to change.

Political centralism sees a wide array of choices within the system. It's basic premises is that the political
culture is the voice of the people as indicated by public
opinion polls. America is a centralist country with strong
roots in some form of private enterprise. Schumpeter,
Galbraith, and lately Scannon & Watenburg maintain that
there are countervailing blocks of organized interest. At
the worst, the political mass are semi-soverign but are
periodically organized around issues and exert their will
through selected lobbyists as well as political representatives. Issues are the important facets of the body
politic rather than ideology. 5

From the readings of the futurist literature, it appears that for the most part trend line projections of super-industrialism see a countervailing Keynisan system.

As Scannon, political analyst, and author of the best seller the Real Majority, notes:

Perhaps this isn't what futurists should say, but I don't see great changes in the next generation in our political system. I don't see it in our institutions. I don't see it in the pattern of the population that supports the institutions. I don't see it in the political system itself. That fact of the matter is that unless we predicate very wide changes among the people themselves -- in their attitudes, in their habits, in their ways of going about their affairs -- we won't find changes in any of the institutions or in the politics of

our system. A democracy like ours, in which people are very closely related to the political structure, is one which changes as need is indicated from paople. A great need was indicated in the depth of the depression, and changes were made. But even those changes, and even the changes most illustrative of postwar America -- the explosion of the working class into the middle class and the flight from the land -- are still more or less in the old pattern.

I would say that if one looks at the future for 10, 20, 30 years, one does not see an American politics substantially different from what it is today. By the year 2000, I would think, the flexibility of our system will likely have produced a number of adjustments, a number of ameliorations, and perhaps new ills, but I wouldn't think it will have produced major overriding changes.

The Commission on the Year 2000 sees a future government somewhat similar to our own. However, the Commission does see a slightly more collectivized system, and more integrated to world order. Senator Humphrey sees more future planning. Mesthine sees more choices in this future political system. Madden, Turner, and Shils see the continuing viability of the Keynsian market system.

National and International Political Systems

The actual federal governmental structure appears to be increasingly Presidential and continuation of this trend is expected. 11 Most sources indicate that no changes will occur; the authors sent a brief open ended questionnaire to key figures in Congress who either had a national reputation or represented the Plains area. For the 23 letters sent we enjoy a 25% return rate in which the majority (anonymity was promised) saw no dramatic changes in the trend. 12

Panel studies of national attitudes indicate to these authors that the American electorate is concerned about

environmental issues, wants government as its employer only as a last resort, and (especially the white respondents) sees the city getting worse. Most segments of the public are suspicious of all big entities (government, business, and laobr). It would appear from public opinion responses that no major or radical changes will occur, for the public lacks a consciousness of kind in terms of the direction of the country.

On the international scene, the picture does not always appear to be as tranquil. Trend line assumption sees a world still fraught with conflict. One source sees gradual nuclear military action. Another sees clamity without world government. A third envisions new major struggles over political assets of the ocean.

Hammel sees the communist block having trouble with the future 17 and Fascell projects more conflict in Latin America. 18 However, Brown is optimistic about the Third World. 19 Japan will probably soon be the most affluent nation in the world. 20

A survey of 120 countries shows mailitary expenditures rising faster than either population or gross product. 21

Karnish sees the spread of nuclear weapons to many countries. 22

Given the trend line projections generally accepted, a large military system is projected for this country. 23

Not all agree what the function of the military in this country will be. One camp envisions the military as a protector of world order. 24 Another has questioned the idea of

the military as protector of corporate market and a wasting agent in the United States. 25

However, there are optimistic accounts of a future devoid of violence which extends to nuclear holicost. 26 For this country, trend line analysis projects continuance in war and war related activities. For whatever reasons, America has been engaged in war activities about three-fourths of her history, 27 and it is therefore unlikely that there will be a diminishment of military expenditures. Due to the economic position enjoyed by the United States social and cultural forms which describe "peaceful" societies will not 28 emerge.

### Governmental Form

Trend line projections do not indicate major changes in this area. General national-local relationships have, however, been developed via federal to regional to state-county-local. Political conservatives want to maintain a federal-state relationship. From the early days of Calhoun and his concurrent majoritarianism, a federal-state relationship has given conservative bias to political structures. 29 Liberals have looked to a federal-city-neighborhood relationship which would benefit liberal constituencies. They see state legislation as a parasitic drain upon city revenues. 30

Another question concerns government formations. The urban coalition (minorities, college youth, White liberals, city mayors, downtown merchants) would like to see a more centralizing and yet representative form of city government.

However, Wood has illustrated that multi-government caused by divided and over-lapping jurisdictions is consistent with the ideological retreat from the city. Suburban government can provide more local attention and can cost less because city resources can be tapped. Though a parasitic relationship, it is beneficial to suburbanites.

Trend line projections do not indicate overwhelming changes in governmental form nor in its geographic constituency. Importantly, multi-government has a special benefit for the more affluent and those who have retreated from the city.

# The Urban Crisis vs. the Logic of Metropolitan Growth

The shape of urban form stems from this disagreement. The urban crisis position says that historically America has always had an anti-urban bias. 31 However, we are now becoming an urban nation with natural increases and rapid rural migrations. City hall is dominated by those who have an anti-urban bias. Generally, the merchant community wants only from the city that which will generate profits and city hall is the arm of merchant elite. Local developers have free rein in chewing up even more agricultural land for profit. Systematic planning is overwhelmed by the elite and their ability to generate revenue. Slumless, greenbelt, masstransited cities are also tyrannized by suburban and rural-oriented state legislatures. The suburbs are parasitical, drawing money from the city but not providing a tax base for the city system. Housing legislation, property tax, slum lords,

highway funding all work to cripple the city. Not only has
the urban area suffered from this, but the farm and rural
areas have lost out also. Reformers in the past have proposed Rural Reconstruction Acts, economic floors, taxes based
on income rather than property, the use of private agencies
to stimulate the city, industrial parks, uni-gov, city redevelopment, city-state approaches, model cities, urban renewal, as well as grants to industry. However all this,
which translates in terms of urban growth into redevelopment
and concentration of people and services, has had little
success.

The authors agree with comments made to us by Dr. John McHale, futurist, urbanologist, noted writer, and confidant of R. Buckminister Fuller that urban planning requires fiscal planning. City planners have for years looked longingly eastward toward the green belts, well planned housing, and efficient mass transit systems of western Europe. However, our political and economic system is considerably different. We do not see this change coming in the United States given trend line assumptions. A survey of history of the last 50 years suggests to these writers that there exists a striking viability of a center right coalition which will serve to preserve the economic arrangement. Notwithstanding the rhetoric of reform, income redistribution has not come about. Relative income distribution has remained about the same. Going back to Roosevelt, who ran on the platform of a balanced budget, the "New Deal" quoting Jimmy Burns was "to

preserve the Seystem." Since that time most successful candidates have been moderate conservative Republicans or Labor Democrats. 32 "Liberals" have been few and have lost overwhelming at the pools. We include here both Stevenson and McGovern. 33 Generally, the United States is more conservative than her industrial counterparts. This reflects in the formation of cities and city life. Sprawl and spread are in the cards for American cities.

If one contemplates the coalitions which will be formed in the future, one is impressed with the probability of emerging conservative coalition. The strength of it should be noted in terms of its power. (Even during the liberal years of the 1960's.) The Congressional Quarterly notes that the conservative coalition (conservative Republicans and Southern Democrats) won on the average between 70-80% of the issues raised. Also to be considered are the number of other issues that were not reported out of committee. 34

One must also consider material contained in two recent books of Kevin Phillips and Scannon & Watenburg. Phillips argues that there is an emerging coalition of suburbanites, Whites, the Business community, the plains, south, and southwest, and religious fundamentalists which insure a Republican majority. Unfortunately for Phillips, it would appear to these authors that if he would have said a Presidential majority he would have been more correct. From our perspective, the moderate conservative of either party, a Connally, Jackson, or Humphrey would have a good chance of winning, if

the majority is not spoiled by an even more conservative candidate like Wallace. If one reads Scammon & Watenburg, they seem extremely defensive. A liberal should fudge the issues they say, playing down the social issues and play up the economic ones. The economic issues to them go along the "New Deal" variety. What to these authors adds up is a continuance of a conservative coalition or an increase in it. "New Politics" appears only to be viable in local elections, and importantly the "New Politics" wanted to drop the south but add the suburban liberal Republicans. However, the coalition assumes the inclusion of the laborite and the southern and eastern community. They now appear to be on the right.

What this brings us to, given a trend line projection orientation, is that Banfield's logic of metropolitan growth appears to be what will occur in terms of urban form.

Banfield indicates that every city administration has at least three elements which narrows their base of choice.

1) population increase, 2) technology and its impact on jobs and transportation, 3) economic, the distribution of wealth and income and its impact on new housing.

Accordingly Banfield asks, "why urban crisis?" "Urban crisis" to that author is a definition conjured up by middle class liberals. Urban decay is really just working class and lower class life style. Middle class define environment desthetically and the working classes define it functionally.

So the classes (middle vs. the working class and poor) don't mix. As the lower classes and working classes spread, the middle classes and upper classes move further away from the central city.

Spread was also encouraged by mechanical refrigeration and canned food which reduced neighborhood centers of boarding houses and resturants. The combustion automobile along with highway legislation allowed the middle classes and working classes to flee. Also, FHA financed new homes rewarded the middle classes to flee to the suburbs. The GI bill, the GI loan, and tract housing also helped encourage spread. After world War II, due to farm foreclosures, there was a rapid influx of Blacks, Southern whites, and unskilled workers into the city. This increased theflight of the middle class to the suburbs. So once the middle class left the central city (except for the diehards and the gold coast area) they return by car to work and they perceive an urban crisis. According to Banfield, today in the city there is actually less sub-standard housing and fewer poor.

So the reforms (public housing, urban renewal, ZPG, services strategies for the poor) all appear likely to be doomed. The living city and corporate provate enterprise do not compliment each other. And Sociologists, urban-clogists, city planners are all likely to be suspect for some liberal reforms that have back fired. The urban co-alition is shaky, because to some Blacks, urban reform may

mean urban renewal which may also nean "Negro removal."

Suburbanites are antagonistic or apathetic and the southern and eastern European community sees urban reform as something to benefit Blacks. Categorically to the southern and eastern European the vision of their world according to Michael Novak is the "good Life" (local neighborhood and parish issues) rather than the "good world" which is the larger societal milieu. Therefore, in terms of the urban community, one source notes:

Urban experts no longer feel, as they did twenty years ago, that drastic interference with the urban system can correct a drastic problem. For example, the programs to eliminate slums not only failed to achieve their goals but created other problems through the relocation of the poor.

The truth is, everybody is tired of cities.
"We could build great cities today, but I don't think
we have the will to do it... 35

### FOOTNOTE3

1 William Ebbenstin et al., American Democracy in World Perspective (New York: Farper & Row, 1970).

As an overview, please see the following sources: Carl H. Horning, "Power and Sociological Stratifi-cation," Sociological Quarterly, 12:3-14, Winter, 1971.

Harold M. Hodges, Jr., Social Stratification, Class in America, (Cambridge: Schenkman Publishing, 1964).

Donald W. Calhoun, Social Science in an Age of Change

(New York: Harper & Row, 1971).

W. J. McKenzie, Politics and Social Science (Baltimore: Penguin Books, 1967).

Henry Akin, Age of Ideology (New American Library,

Joseph Monsen and Mark W. Cannon, The Makers of Public Policy (New York: McGraw Hill, 1965).

Richard R. Fagen, Politics in Communication (Boston:

Little, Brown, 1966).

Joseph Gusfield, Protest, Reform, and Revolt -- A Reader in Social Movements (New York: John Wylet & Sons,

Calvin J. Larson and Philo T. Washburn, Power, Participation, and Ideology (New York: David McKay Co., 1969). Eric A. Mordlinger (ed.), Politics in Society --Studies in a Comparative Political Sociology (Inglewood Cliffs: Prentice-Hall, 1970).

John Kunkl, Society of Economic Growth: Upheaval of Perspective by Social Change (London: Oxford University Press, 1970).

Bernard C. Hennessy, Public Opinion (Belmont, Cali-

fornia: Duxbury Press, 1970).

Edward C. Dreyer and Walter A. Rosenbaum, Political Opinion and Behavior -- Essays and Studies (Belmont, California: Wadsworth Publications, 1970).

Reo M. Christiansen and Robert O. McWilliams, Voices

of the People (New York: McGraw Hill, 1967).

Richard Appelbaum, Series of Social Change (Chicago:

Markham Publishing, 1970).

3Max Eastman, Reflections on the Failure of Socialism (San Diego: Viewpoint Books, 1955).

Senator Strom Thurmond, The Faith we Have not Kept (San Diego: Viewpoint Books, 1968).

Barry Goldwater, The Conscience of a Majority (Inglewood Cliffs: Prentice-Hall, 1970).

J. Evetts Haley, A Texan Looks at Lyndon -- A Study in Illegitimate Power (Canyon, Texas: Palo Duro Press, 1964).

Phyllis Schafely, A Choice, Not An Echo (Alton, Illinois: Pere Marquette Press, 1964).

Herb Streith and Gordon King, Action for Americans --The Liberty Amendment (Los Angeles: Operation America, 1965).

Gary Aaron, Mone Dare Call it Conspiracy (Rossmoore.

California: Concord Press. 1971).

Barry Goldwater, Conscience of a Conservative (New York: Hillman Books, 1960).

John A. Stormer, Mone Dare Call it Treason (Missouri:

Liberty Bell Press, 1964).

Dr. Fred Schwartz, You Can Trust the Communists to do Exactly as they say (Inglewood Cliffs: Prentice-Hall, 1960). Jose Ortega y Gasset, The Revolt of the Masses (New York: W. A. Norton & Co., 1960).

Barry Goldwater, Where I Stand (New York: McGraw-Hill,

(New York: Bantam Books, 1969).

Burt Cochran, Adlai Stevenson (New York: Funk &

Wagnells, 1968).

James Davis (ed.), When Hen Will Vote and Why -- A Reader in Political Violence and Revolution (New York: Free Press, 1971).

MAO Tse Tung (ed.), Quotations From Chairman MAO-TSE-TUNG

(New York: Bantam Books, 1967).

Herbert E. Robb and Raymond Sobel (eds.), From Left to Right -- Readings on the Social Political Spectrum (Benziger,

Richard C. Edwards, Michael Reich, and Thomas E. Weisskopf, The Radical Analysis of American Society -- The Capitalist System (New Jersey: Prentice-Hall, 1972). George Kirstein, The Rich -- Are They Different?

(Boston: Houghton Mifflin, 1968).

Cy Kahn, How People Get Power (New York: McGraw Hill,

Morton Mintz and Julius Cohen, America Incorporated (New York: Dell Books, 1971).

Frederick L. Bender (ed.), Karl Marx -- The Fssential

Ernest Mandel, Europe vs America: Contradictions of Imperialism (New York: Modern Reader, 1970).

Roderick Aya and Morman Miller, The New American

Revolution (New York: Free Press, 1971).

Jules Henry, Culture Against Man (New York: Vintage Books, 1963).

Angus Black, A Radical's Guide to Economic Reality

(New York: Holt, Rinehart, dinston, 1970).

Paul M. Sweezy and Harry Magdoff, The Dynamics of U.S. Capitalism -- Corporate Structure, Inflation, Credit, Gold, and the Dollar (New York: Monthly Review Press, 1972).

Leo Hamaliam, Frederick R. Farl, and Thomas Y. Crowell (eds.), The Radical Vision -- Essays for the Seventies

(New York, 1970).

Matthew F. Stolz, Politics of the New Left (Beverly

Hills: Glencoe Press Insight Series, 1971).

John Kenneth Galbraith, American Capitalism (Boston: Houghton Mifflin Co., 1956).

Hugh A. Bone, American Politics and the Party System (New York, McGraw Hill, 1971).

Frederick Pohl, Practical Politics 1972 (New York: Valentine Books, 1971).

William Ebenstein, et al., American Democracy in World

Perspective (New York: Harper & Row, 1970).

Movement for a New Congress, Vote Power -- The Official Activist Campaigner's Handbook (Englewood Cliffs: Prentice-Hall, 1970).

Samuel Lubell, The Future of American Politics (New York: Doubleday Anchor Books, 1956).

6Richard Scammon, "The Future of the American Political

System," The Futurist, August, 1969, pp. 97-100.

7 Joseph F. Coates, "The Future of the U. S. Government," (editor of The Future of the U. S. Government: Toward the Year 2000 by Harvey S. Perloff), The Futurist, June, 1972, pp. 104-108.

J. Thompson (review), "The Future of the U. S. Govern-"Harper's, 243:89-91, August, 1971.

8Herbert H. Humphrey, "Whither We are Tending: An Effort to Make the U. S. Government Look to the Future." The Futurist, December, 1972, pp. 236-238.

9 Immanuel Mesthene, "Is Technology Destroying Human

Values?" The Futurist, August, 1969, pp. 94-95.

Daniel P. Moynihan, "Goal Systems and Hidden Policy,"

The Futurist, August, 1970, pp. 119-120.

10Robert C. Turner, "The Long Future of the Economy,"

Business Horizons, XIII:3, June, 1970.

Carl H. Madden, "Business and the Future," The Futurist, December, 1969.

Edward Shils, "Notable and Quotable," WSJ, May 4, 1973.

11 Richard E. Neustadt, Presidential Power (New York,
New American Library, 1964).

Rexford G. Tugwell, "The Historians and the Presidency --An Essay Review, " Political Science Quarterly, 86:2:183-204. June, 1971.

Louis W. Konig, "The President Needs a New Kind of Power," The Annals of the American Academy of Political and Social Science, 71-82, Spring, 1970.

"The Rise of Congress in Foreign Policy Making," The Annals of the American Academy of Political and Social Science, September, 1971, pp. 61-70.

Milton C. Cummings, Jr., Congressman and the Electorate (New York: Free Press, 1966).

Donald R. Matthews, U. S. Senators in the World

(New York: Vintage Books, 1960).

Ralph Mader (for Congressional Project) Who Runs

Congress? (New York: Grossman Books, 1972).

Donald Herzberg and Gerald Pomper, Party Reform in the Future: American Party Politics, Essays and Readings (New York, Holt, Rinehart, Winston, 1966).

David B. Truman, The Congress and America's Future

(Prentice-Hall, 1972).

12A questionnaire was sent to the number of senators and representatives as listed in the manuscript. These are the people who were contacted; of which the response as noted before, not all returned the questionnaires and made comments: Senators Walter Mondale, Hubert Humphrey, George McGovern. Richard Clark, Harold Hughes, Thomas Sagleton, Carl T. Curtis, Roman L. Hruska, Stuart Symington, Robert Dole, Edward M. Kennedy, Hugh Scott, Mike Mansfield, Robert Byrd, Strom Thurmond, John Tower, Barry Goldwater. Representatives Donald Regal, Bella Abzug, Carl Albert, John Y. McCollister, Gerald

Ward, and William Scherle.

13 Joseph R. Goeke, "America's Future as Reflected in Public Opinion Trends," The Futurist, April, 1971, pp. 79-

14 Daniel Bell, et al. (Panel) "working Sesson Two:

Four Futures, " Commentary, 1970.
15Frank Snowden Hopkins, "Courage Toward Utopia," The Puturist, April, 1970, pp. 65-66.

16" Who Will Own the Sea?" The Futurist, June, 1968,

pp. 41-44.

17 Ralph Hamil, "The Cloudy Future of Communism," The

Futurist, December, 1970, p. 213.

18 Dante B. Fascell, "Latin America and the U. S.: New Directions for American Policy," The Futurist, June, 1970,

p. 9519Frank Snowden Hopkins, "The Underdeveloped Countries Now Dare to Hope," The Futurist, June, 1970, pp. 103-104. 20 James C. Abegglen, "The Economic Growth of Japan."

Scientific American, March, 1970.

21 Archibald Alexander, "The Cost of World Armaments," Scientific American, October, 1969.

22Arnold Kramish, "Some Long-Range Implications of Nuclear Proliferation," The Futurist, December, 1968, p. 122. 23For an overview of American military system, please see: Kurt Land, Military Institutions and the Sociology of

War | (Beverly Hills: Sage Publications, 1972).

Charles C. Moskos, Jr., The American Enlisted Man

(New York, 1970).
24 Zevedei Barba, Democracy and Dictatorship (New York:

Gold Press, 1956).

Kurt London, The Permanent Crisis (Toronto: Blaizedell Publications, 1968).

J. V. Wolfe and John Erickson (eds.), The Armed Services and Society (Edinburg: Edinburg University Press, 1971).

25 John C. Bennett, Nuclear Weapons and the Conflict of Consciousness (New York: Charles Scribners Sons, 1962).

Lartin B. Hickman, Military in American Society (California: Glencoe Press Insight Series, 1971).

C. Wright Mills, The Causes of World War III (Valentine Books, 1960).

Seymour Melman, Pentagon Capitalism -- The Political Economy of War (New York: McGraw Hill, 1970).

26 Arthur I. Waskow, 'Creating the Future in the Present," The Futurist, August, 1968, pp. 75-76.

V. Bastuk, "The Impact of Technology in the Next

Decades," Sprint, 14:1:17-42, 1970.

Arthur I. Waskow, "The Berlin Crisis of 1999," The Futurist, April, 1967, pp. 24-27.

27 Hobert R. Alfword et al., Society Today (Delmar,

California: CRM Publishers, 1972), see Chapter 33.

28 Mathen Melko, "Discovering Peace," Peace Research,
December, 1971, Vol. 3, No. 12, pp. 16-17.

29Hennessy, Public Opinion and Political Behavior, op cit. 30 Robert F. Kennedy, Seek A Newer World (New York)

Bantam Books, 1968).

John V. Lindsay, The City (New American Library, W. W. Norton & Co., 1970).

32 Morton Mintz and Jerry S. Cohen, America Inc. Who Owns and Operates the United States (New York: Dell, 1972).

33However, Cochran has raised issue with Stevenson as a

"liberal."

Burt Cochran, Adlai Stevenson (New York: Funk & Wagnalls, 1969).

34"The Conservative Coalition," Congressional Quarterly,

1971, p. 81. 35 Earvey S. Perloff, ed., The Future of the United States Government: Toward the Year 2000 (Braziller Publisher, 1969).

Chapter Five

The Economic Outlook

### The Economic Outlook

## Move Toward Service Industries

The social system presented earlier is defined as "super-industrial," It has a strong basis in economics. The key to super-industrialism is the change of focus of the means of production from basis and secondary industries to tertiary (service) and quaternary industries (government, research, and development, etc.). This projection includes the extension of a Keynsian market system as described by Galbraith and others. 2

The land force would be better educated, and work less often, and enjoy more employee benefits. The profile of the labor force includes heavy addition in the service and quaternary areas. As Klemme notes:

By the year 2000 A.D.from the standpoint of the industrial developer, the most important changes will be found in the structural pattern of employment and the relative and absolute changes which will occur in the last third of this century.

Basic employment (manufacturing, agricultural, and mining), which is the employment generating basis on which all other employment is supported, will grow by only about 4 million jobs between 1965 and the year 2000. Manufacturing will increase by some 8 million jobs, but this gain will be offset by decreasing employment in agricultural and mining. The basic supporting industries (transportation, utilities, construction, etc.), will gain approximately 5.7 million jobs but will decline from 11.4% of the total employment in 1965 to approximately 10.2% of total employment in the year 2000. The almost exploding employment potential in the fields of service, trade, and finance will continue through the year 2000. Employment in these fields will rise from approximately 26.8 million jobs in 1965 to 63.4 million jobs in the year 2000. This will account for nearly 60% of the total growth of employment in the last third of the century.4

Another source indicates that by 2068 service sector labor will reach nearly 75% (from 55% currently) while manufacturing labor will account for only about 20%. 5

# Consumption and Productivity

Higher total consumption is predicted not only because of the national increase in population, but because of what Toffler calls the economics of impermanence. He notes that new products which are produced cheaply (and thus easily replaced) will win consumer acceptance over those which are guild or custom repair oriented. Rentalism is also thought to increase.

Because of advances in technology, labor productivity should increase by about 3% per year. The average work week is projected to be something like 30-35 hours per week.

Daily means of transaction may be totally carried out by credit card. 10 Capital investment will continue to increase because consumer outlays will continue to increase as will government expenditures. 11 The overview then of this system is one of increasing centralization of both government and corporate enterprise to the point where the resource management of the
two entities become indistinguishable. Service-related
industry should be dominant and consumption should be
increased.

## A Super-Industrial Economy and Urban Growth

How does economics translate into urban growth? Turner, in dealing with the future of urban growth, notes:

People are shifting out of primary industries

(agricultural and mining) into secondary industries (manufacturing and goods processing), and out of these into tertiary industries (the service industries such as retailing, transportation, finance, entertainment, and so on), and finally into quaternary industries (education, government, and the arts). These occupational shifts have produced geographical shifts: people moved from the farm into the giant metropolitan areas, then out of them to the periphery. I predict that the

geographic trend of the next several decades will be away from the metropolis to locations beyond the periphery, along arterial highways, thus creating strip cities hundreds of miles long, and to smaller cities, some of them in the strip cities, others only a short hop by high-speed transportation from the metropolitan areas. 12

Land economics is concerned with key concepts that

deal with man-to-land relationships. First "urban functions" considers the historical and economic beginnings

of an area. Included under these concepts are the following concerns: 1) what is the basic means of production for an area? 2) what kinds of tax revenues are
generated? 3) is there available low land for transportation and poor land for industry in addition to scenic
areas for tourism and use by the general populace? 4) what
is the relationship between heartland and hinterland (such
as nearness to minerals and succeeding stages of production)?

5) what is strategic about the location, such as its
nearness to agriculture, national markets, and other trade
centers? 6) what facilities help to bring about an inward

flow of materials and an outward flow of finished products?

7) can an area attract "people to services" (e.g., aged, defense industries, government, education, etc.). 13

Generally basic theories within this area include "trade theory" (which is concerned with supply of land, labor, and capital), "location theory" (dealing with location of firms and groups of firms), "staple theory" (which looks at migratory processes and economic benefits), and "economic base theory" (which analyzes the basic means of production of the area). 14 Though there are other theories, we will draw upon only these and in a collective fashion. 15

Given the super-industrial economy, we should see the centripetal pull of the downtown core still diminishing.

Though there will be an additional office space added to the downtown in some areas, we see the core as a commuter-consumer related retail area continuing to diminish in most downtown sectors. 16

On balance we see a "competitive shift" from core to suburban tracts, keenly spurred on by the interstate and other resource endowments. One of the struggles in the future will be between relatively cheap suburban land or

redeveloped industrial parks. Given the nature of the super-industrial state, suburban expansion appears to remain predominate.

The exurban industrial/commercial ring will demand more area in order to provide space for parking. As we will note later, the mass transit system will be an important element but not a dominant one. Even projections for the 1980's envision more commercial acreage because of the new shopping areas which require between 3 and 5 times the space needed by World War II retail outlets. 17

The labor profile should remain similar to present conditions, with the exception of increases in tertiary and quaternary areas. This does not mean that the relationship in any way will always be direct. Predictions translate into higher percentages of employment as follows: wholesale and retail trade, finance, personal services, entertainment and recreation, and public administration. 18

We project an unemployment rate of between 4% and 6% as presently measured. Industries with a high nature of

employee turnover need a ready pool of labor resources to draw upon. 19

As an overview then, the super-industrial state will, on the economic level, promote scatteration.

when new industrial locations come about, the exurban ring and others in the nodal region will be more likely to locate in the suburbs because of the following reasons:

1) relatively cheaper land, 2) not as amenable to sources of organized labor, 3) middle income labor base, 4) less amenable to political activist groups, and environmental groups, 5) lower tax base. 20

Incidentally, when we indicate suburban, we do not entirely mean growth from the core radiating out to the suburbs. We also see growth coming from the smaller towns to the city. This is especially true with service-related industries, those not directly dependent upon the major mass transportation entities.

The super-industrial economy will be a fluid one, but continually regulated.

### A Planned City

What would happen to urban form if this general evolutionary drift did not come about? What would happen to the city structure given a post-Keynsian or social market economy? Productivity would diminish slightly. Unemployment would be eradicated, and the tax load would be increased. Few or no ghettos would exist, the upper classes would take on a lowered profile. Much of the real national income would be channeled into the public sector. The military and armaments might also be rechanneled. Income redistribution would occur. Housing starts would be channeled through a master plan developed by city planners.

Greenbelts, bike paths, and public housing would become more abundant. The growth in technology would diminish somewhat while human services areas would increase. Both the central core and exurban shopping areas would be redeveloped. Population explosion may be less tolerated due to the change in the terrain of the city.

Cottage development would take precedence over the sprawling ranch and split-level home. Striking observable differences among classes would be diminished.

Private enterprise would be a viable subculture especially predominate in the service sector. Economic growth would be regulated by a combination of consumer wishes and central planning.

Most cities would less likely grow outward, but grow through redevelopment and vertical expansion. New growth, manpower, and the tax base would likely increase in heartland and hinterland.

## The Restoration City

What would happen within the city if the general drift would return to a prior economic system?

For those who maximize their opportunities and economic freedoms, there would be strikingly attractive
suburban homes, perhaps even large manors. The upper
income and upper middle income (which now account for
11% to 13% of the population) would be somewhat reduced
but those who remain would have increased real income.

The changes in GNP would vary strikingly from year to year.

Real dramatic prosperity would follow years of economic depression, and the unemployment rate would at times be as high as a quarter of the employment force.

Those institutions now directed by the public sector would tend to revert to voluntary associations. Government would be responsible for only minimum tasks such as fire protection, highway maintenance, and sanitation services.

Urban horizontal expansion would be appropriate.

Changes within the city would become more pronounced. By custom, populated impaction might become more noted in low income areas. Depending upon the era, workers would come to rely more heavily upon mass transportation and would be more likely to gravitate to the core of the city.

With social services reverted to voluntary associations, the church and religious life would probably take on a new significance. Income distribution would likely drift upward. Productivity and efficiency would increase dramatically. However, city expansion would not be as

great because of impaction of the poor, and the job-time-locale consideration of the workers.  $^{22}$ 

### Chapter Five

#### Footnotes

1"America's Next 30 Years: Business and the Future," (Report from the U.S. Chamber of Commerce Council on Trends and Perspectives, Economic Analysis and Study Group, Washington, D.C., 1970).

Dennis Gabor, Inventing the Future, 1963.

George H. Daniels, "The Big Questions in the History of American Technology," <u>Technology & Culture</u>, 11:1:1-21, January, 1970.

<sup>2</sup>John K. Galbraith, <u>The Affluent Society</u> (Boston: Houghton, Mifflin, 1958).

Joseph A. Schumpeter, Capitalism, Socialism, and Democracy, (New York: Harper & Row, 1950).

John K. Galbraith, The New Industrial State (Boston: Houghton, Mifflin, 1967).

Joseph C. Goulden, Monopoly (New York: Pocket Books, revised edition, 1970).

Fritz Heimann, The Future of Foundations (Harcourt, Janovitch, &Brace, 1973).

3 Ibid.

Randal T. Klemme, "Industrial Development -- 2000 A.D."
The Futurist, February, 1968.

5 Ibid.

<sup>6</sup>J. Jamison, "What Business Should Do Now to Anticipate the Future," <u>The Futurist</u>, February, 1969. Toffler, op. cit., p. 56.

Robert C. Turner, "The Long Future of the Economy," Business Horizons, XIII:3, June, 1970.

8Ibid.

9"A Look at the Great Economy of 1985," Business Week, December 18, 1971, pp. 84-85.

10For further information contact Liesa Bing, The Diebold Group Inc., 430 Park Avenue, New York, N.Y. 10022

11"A Look at the Great Economy of 1985," op.cit.

Klemme, op.cit.

"The Road to a 200-Billion Budget," U.S. News & World Report, February 16, 1970, pp. 64-65.

William F. Butler (ed.), How Business Economists Forecast (New York: Prentice-Hall, 1966).

12Turner, op.cit.

Richard M. Hurd, <u>Principles of City Land Values</u>
(New York: The Record Studies, 1903).

Stanley McMichael and Robert F. Bingham, <u>City Growth</u>
<u>Essentials</u> (Cleveland: Stanley McMichael Publishing Organization, 1928).

Herbert D. Dorau and Albert G. Hinman, <u>Urban Land</u> <u>Economics</u> (New York: The MacMillan Co., 1928).

D.J. Reynolds, Economics, Town Planning, and Traffic (London: Institute of Economic Affairs, 1966).

Brian J.S. Berry, <u>Strategies</u>, <u>Models</u>, and <u>Economic</u>

Theories of <u>Development in Rural Regions</u> (Agricultural Economic Report No. 127), <u>Department of Agriculture</u>, Washington, D.C., U.S. Government Printing Office, 1967).

- 15 Others include input output analysis, econometric models, simulation models, and spatial programming.
- 16 See Figure 85 of Brian J.S.Berry and Frank E. Horton, <u>Geographic Perspectives on Urban Systems</u> (Englewood Cliffs: Prentice-Hall, 1970).
- 17 Alden Aust, <u>Community Renewal Program</u>, City Planning Department (Locally published), 1966.
- 18For current employment figures, please see MAPA Economic Base Survey, City Planning Board (Locally published), June 30, 1970.
- Metropolitan Omaha, Nebraska, Council Bluffs, Iowa, Plan of Study, Corps of Engineers (Locally published), 1972.
- Edwin D. Goldfield, Congressional District Data Book (Washington, D.C.: U S Government Planning Office, 1972).
- Edwin D. Goldfield, City and County Data Book (Wash- 'ington, D.C., 1972).
- 19 See current rate in William Lerner <u>Statistical</u> <u>Abstract of the United States</u> (Washington, D.C.: U.S. Government Printing Office, 1972).
- 20 Irving Kristol, "Is New York City in a Crisis,"

  Social Profile USA Today (New York; Van Nostrand Reinhold Company, 1970).
- Por an exhaustive overview in this area, please see Harry W. Laidler's <u>History of Socialism</u> (New York: Thomas Y. Crowell, 1968).
- Though there is plenty of literature on the political Right, few list budgets. However, please see the budget list in Lloyd G. Herbstreith and Gordon Van B. King's Actions for Americans, The Liberty Amendment (Los Angeles: Operation

America, 1965). To quote from the frontice page of the book:

"The Liberty Amendment states that the Federal Government shall not operate business-type activities unless . they are specifically authorized by the Constitution.

"It provides a three-year period for selling or liquidating the more than 700 business-type enterprises presently operated by the Government without constitutional authority. Sale of these enterprises will bring in enough money to reduce the national debt by at least twenty percent. Annual budget spending by the Government will be reduced by more than fifty percent. Revenue from excise taxes on goods and services, and on corporation incomes, will increase at least twenty percent, without increase of tax rates.

"This means that the annual revenue collected from the Federal Personal Income and Withholding Tax, the Federal Estate Tax, and the Federal Gift Tax, will not be needed. So the Liberty Amendment will stop these three types of taxes, at the end of the three-year period." Chapter Six

The Sociology of Housing

# Introduction

In the early 1960's, Pete Seeger popularized a song about little "boxes," pink ones, yellow ones and the like. The message of the song was that small suburban-tract housing was monotonous, undifferentiated, and a general blight to the natural terrain. Suburban-tract housing became especially popular after World War II, with the financing of new mortgages by the Federal Government. For many it was the ideal escape from the city. Though outwardly attractive to the consumer; because of its singularity, treelessness, and lack of parks and originality, tract housing of Levitt variety was not favored by planners and artisans. Importantly, the tract housing was responsible for large amounts of land consumption.

At the same time, the inner-metropolitan area was losing its middle-class tax base to lower income groups which began to dominate the city in terms of population. Housing for the underclasses became popularized by the institutional high rise.

'Though public housing appeared to be viable for the cyclical poor, the structural poor appeared harder to house. Institutional high rise systems presented problems for those with a high number of children per completed family. Beatings, muggings, and rapes occurred in the hallways of high rises. Total tenant-run management ended up in such examples as Prewitt-Igott where the buildings were too expensive to repair. (High rise systems did however seem viable for the aged.)

Tenant-city hall struggles ensued between Jesse Grey
(NTO) and other organizations for renter rights.

Presently, rebuilding and refurbishing individual homes has not met with the success one would like.

As an overview, where the land developer and realtor initiate new city expansion, that expansion is generally outward.

In a land development system, a "trickle theory" in terms of housing appears to emerge. The typical middle income area begins new, treeless, and expensive. Though not all suburbs are alike, some similarities generally

appear. The typical neighborhood houses families as opposed to singles, or aged couples. As time continues, the neighborhood slowly deteriorates though at times it may last much longer due to restoration processes or location close to attractive consumer space areas. But at some point, the change comes and the single-owneroccupied home is, sold to the next lower income group and in time to yet another income group. In the meantime, the demographic profile begins change by ethnicity, income, age variation, and family life style. Generally, single dwellings become multiple ones and owners become predominated by rentiers. Because the regressive nature of the property structure works against tax improvements, others (sometimes called "slum lords") find that more revenue can be generated by reducing the internal space of the structure and providing housing for more families. In the end, the buildings (sometimes called "hollow shells") are condemned and revert back to the city. Categorically, it provides a base of land disuse for the city as a tax burden.

On the other hand, suburban growth is uninterrupted unless there is a physiographic impediment. Suburbanites have many benefits. In fact the city should be looked upon as a support system to the suburb. Generally, the tax base is lower, suburbanites can utilize the advantages of the city while maintaining and insuring homogeneity of like social backgrounds in the neighborhoods. Over and above that, the local suburban governments can be more immediate and responsive to their needs, and importantly, as annexation occurs the county institutions have reduced work loads and are, therefore, able to provide better services e.g., street repair and snow removal.

Not all suburbs grow and become old in the same way.

In more affluent suburbs the process is much slower with,
in many instances, an outcome resembling a gold coast
phenomena. Large old homes become retail shops, stores,
and apartments. A suburb that is first inhabited by working class or low income residents has a growth life which
is much shorter.

The tract house, along with the social and economic phenomena described earlier by Banfield (along with the automobile), all give rise to urban scatteration. It would appear that individual city reform programs only delay this process.

### The Overview

Where are we today in terms of housing? As a nation, the U.S. Department of Commerce indicates:

- For the first time in U.S. history, there are more housing units in the suburbs than in the central cities of our metropolitan areas or in the non-metropolitan areas.
- Home ownership grew proportionately more among blacks than among whites between 1960 and 1970.
- There was a great increase in the number of mobile homes and trailers during the 1960's.
- The number of apartments increased proportionately more than single family homes during 1960-1970.

- The value of our homes went up greatly from 1960-1970.
- There was a similarly large increase in the amount paid for renting homes.
- Homes in 1970 were less crowded than in 1960
   and earlier.
- More homes had basic plumbing in 1970 than ever before.<sup>2</sup>

It would appear for Americans that there are now more renters than home owners, a phenomena which appears contradictory. But the important point is that there is an increase in both because there is an undoubling of households with the decline of the extended family to a neo-local one. This may also account for the statistics which show that housing is less crowded.

Brown<sup>3</sup> thinks we will see more single neo-local dwellings.

# Future Housing: Super Industrial Trend Line Assumptions

What will housing be like in the future if there are no dramatic social and economic changes?

First, it is possible that the most dramatic aspect of housing will be in the interior. Northwestern Bell<sup>4</sup> projects picture phones for the Minneapolis area by 1976-78. Video screens, flexible walls, functional furniture, increases in the use of plastics, and increases in mobility between homes (dependent upon the age-life-cycles) are likely to occur.<sup>5</sup>

However, neither the exterior nor the type of construction will be dramatically different. Though the labor union movement is small in the United States, the skill trades are extremely strong. As is the local developer, so also is the trade union jeopardized by modular housing.

Given super industrial trends, it is difficult to predict the future of modular housing.

Clark<sup>6</sup> sees that by the end of the 1980's, modular concepts will have beaten custom building except for the luxury market. He also sees the cost of housing encouraging smaller homes and less yard space.

Jamison Moore 7 indicates:

New housing and urban redevelopment will create such a large market for the home construction industry that it will finally have to innovate and make changes to keep up.

Fowler also sees such a demand for new construction that it will require mass production.

On the other hand, U.S. News and World Report<sup>9</sup> is not as optimistic, and recent court battles have favored custom builders.

Modular systems appear to be the key to solving American housing problems according to an ABC documentary "The Building Innovators." If modular systems do not succeed there would be major increases in suburban apartments, condominiums, and mobile homes. In some respects the mobile home of the future may be the prefab and the tract house of the World War II era.

From the authors' perspective the degree of spread is dependent upon modularism. If modularism succeeds, we will see higher land consumption than if it does not. On the other hand, unplanned mobile homes and apartment dwellings may take a little less land, but they also have less space devoted to green areas.

Over and above the modular issue is the appearance that the construction industry will centralize and the majority of new housing starts will occur in the southern half of the nation. 11

# Housing and Urban Form

Generally, single dwelling and its complementary yard space is a large consumer of land. Every major new design tries to incorporate privacy and territoriality into the space including, if possible, some outside consumer space.

In terms of the environment, housing consumes 35% of the energy production. 12

Data on consumer preference and land use indicates that income is still the major variable. 13 Indicative of this is what urban geographers call individual characteristics of consumer housing preference. Besides income, stage in life style, life style preference, and attitudes toward journey to work are important. From their information housing changes are dependent upon price, type of house, type of neighbors, community institutions, and location with respect to a job.

upon the types of land control, multi-dwellings and condominiums can provide less land use and more green area. It
appears to work when territorial and open-air social space
is built into the high rise along with nearby (2 blocks
or less) green area parks. In this instance land consumption is reduced. Secondly, land consumption is reduced if
cottages are erected where doors do not face each other
and there is a heavy concentration of wood and underbrush,
and life cycle segregation can occur. Third, in high rise,
recent evidence indicates that individuals recude crime
and have a higher identity with the project when a small
number of doors face each other.

As an overview, man-land relationships are dependent upon the degree of single dwelling houses and the spatial distance from each other.

# An Arcological Society\*

What if there is a dramatic change in the social-

<sup>\*</sup>The authors have attempted to portray each system based on the assumptions and value judgments of each system.

economic spectrum akin to a Green Revolution (described by Reich)? What effect would this have on housing?

First, megapolitan growth would be redistributed to urban core, new towns, and rural areas. Growth itself would be controlled through governmental agencies via an overall master plan. Zoning, housing codes, and other features would fit with existing established statutes.

Soleri<sup>14</sup> suggests an arcology system where man and land are viewed as working in harmony. Large numbers of people would distribute themselves in building systems adapted for territory and open-air social space. Greenbelts, woods, bike paths, and man-made lakes would be developed within the urban core, yet distributed in such a way as to be within short walking distances of the high-rise. Industrial parks would be allocated away from living areas and hidden by foliage, polychromatic ecological colors, and architectural design. Commuting would be accomplished by mobile sidewalk, electric car, and personal rapid transit system.

Though traditional cottage designs, angular forms, and established forms would remain, there would be an emergence of Fuller's geodesic domes. 15

Governmental representation based upon land (e.g. senator from each district) would be changed and representation would be based on occupational categories (e.g. representation for plumbers, teachers, clerks, etc., or what is called a "syndicalist system"). Local government now based on agricultural divisions (county and state) would diminish and there would be a new emergence into super community government, regional government, and federal level government. 16

For a particular area, a super community government

(with corresponding neighborhood city halls) would probably

be responsible for a metropolitan system and surrounding

satellite cities. With an interdependent relationship

with the regional government and the Federal government

in Washington, D.C.

# The Restoration Tract System

On the other hand, what if the system should revert to the previously established system ?

As the economic growth becomes more based on real prosperity of the laissez faire variety than from the "forced" prosperity of Keynesianism, there may be a doubling of households. Living with relatives is an ideal system when one is dependent upon voluntary associations for welfare, medical care, and the like. It is also extremely helpful during depressions.

With the doubling of households, the skilled working class may drift back into the urban core because of economic depression. Housing may become slightly larger and more functional (e.g., the house-frau design system). 17

Flight to the suburbs may be diminished because public law, in terms of open housing, would return to social custom and the individual's ability to use his economic freedoms. These abilities are generally considered along genetic and sexual biology. Housing segregation and selected impaction and concentration of the Third World Community would ensue.

on the whole, the cities' housing, public library systems, schools, etc., would take on a lower profile in terms of esthetic appearance.

For those who through hard work, talent, and inheritance could make it to the top in terms of the "American Dream," housing would be palatial. A neo-gold coastal phenomena would return. Elegant, hand-crafted custom housing could be produced because of much-reduced labor costs.

Generally, the spread that would occur would be less than the super-industrial trends, but more so than an arcological system.

## Conclusion

Housing is a visible element of urban form. Future projections of city life must include this basic element of social organization.

Housing and its spatial distribution is highly related to the total system. It is suggested that the private land development process is most likely to occur in the future.

#### Footnotes

- Henry J. Aaron, Shelter and Subsidies: Who Benefits from Federal Housing Policies?, (Brookings, 1972).
- <sup>2</sup>U. S. Department of Commerce, <u>We</u>, the Americans, Our <u>Homes</u> (Social and Economic Statistic Administration, Bureau of the Census, October 1972).
- 3George H. Brown, "1984...plus 1," The Conference Board Record, Vol. III, No. 12 (December, 1970), pp. 20-24.
- 4Roger Hahn, representative of Northwestern Bell, persona\_interview.
- 5"Glimpse of the Home of Tomorrow, " U.S. News & World <u>Heport</u>, 64:88-90, July 3, 1967.
- 6Lindley H. Clark, Jr., "house on the Move, " Wall Street Journal, May 7, 1973, p. 12.
- 7J. Jamison Moore, "What Business Should Do Now to Anticipate the Future," <u>The Futurist</u>, February, 1969, p. 27.
- <sup>8</sup>Glenn Fowler, "In Decade's Blueprints: 25 Million Homes, 110 New Cities," New York Times, January 11, 1973.
- 9"Wondrous World of 1990; Outlook for Young People."
  U.S. News & World Report, 62:62-66, January 30, 1967.
- 10B. & B. Contruction Company, "The Building Innovations," a personal interview with Roger Brown.
- 11"The Next 65 Years," Financial World, 128:25ff, October 25, 1967.
- 12"The Energy Crisis: Time for Action." Time, May 7, 1973, p. 44ff.

Chapter Seven

Transportation Systems

## Transportation Systems

## Introduction

The beginning of the American city can be visualized as a tightly compacted area where mobility was carried out by foot or animal power. Up to World War II, mass transit systems were major carriers for many in the city. From that time period on, the internal combustion automobile became the vehicle of personal conveyance.

Mortimer gives an overview of the early problems of mass transit systems and why they failed during the war years.

... Some of these financial failures were due to poor planning, inefficient operation or even financial chicanery. There were other failures caused by excessive and unrealistic demands written into transit franchises. Most transit companies were required to remove the snow, not only from their tracks but also from the adjacent traffic lanes of the streets. They were required to sprinkle the streets to keep down

the dust and the blowing of the pollutants. Often
the companies were required to build and operate
lines into undeveloped areas of the community or to
operate all night "Owl" service on lines where the
crews' main problem became loneliness. Since there
were a lot more patrons than stockholders, the companies were convenient whipping boys and holding
the fare down became a favorite vote-getting project
for the local politicians.

While wages and power costs increased with the years, fares remained comparatively constant. When one views the buses, trains, tracks and depots of a modern transit system, it is difficult at first glance to realize that a transit system is a labor intensive system. A quick check of a typical transit system's operating report will show how much money is paid out in wages. 2

Since the war, in most areas support for the system has diminished.

Thus the automobile emerges as the predominate mode of transportation. Data indicates that urban America will use

the auto for even the shortest trips. In many instances, anything beyond three blocks requires the car. 3

## The Automobile

The popularity of the automobile appears to be viable and enduring. Chrysler Corporation predicts boom sales in the early 1980's. Projections for the year 2000 envisioned nearly one car for every adult. Lindsey notes that the auto is expected to remain the country's number one means of travel.

Why the success of the car? Keeffer suggests:

What is there in the automobile's hold on the public that could be usefully incorporated in transit planning?

- An origin-to-destination method of transportation.
- Normally Available at all times.
- 3. Goes in any direction at the whim of the driver.
- 4. Doesn't have to stop and pick up others.
- Provides privacy and reasonable safeguards against the annoyance of others.

No other method of transportation since the horse and buggy has afforded such advantages to the individual driver and his family. 7

road and highway system, built around the needs of the car has emerged. A motorist can drive 1,800 miles non-stop, except for toll booths and gasoline fills, from Massachusetts to South Dakota. The interstate highway system is now about two-thirds complete. Projections are to link every population center of 50,000 or more. Even land around the interstate has increased in value as well as productivity. 10

However, if the individual auto survives in the future, it may change, as indicated by <a href="Financial World">Financial World</a>:

...The transportation and travel field will be one of the giants in the U.S. Economy over the next 65 years. The motorist probably will be driving a car with a different transmission system, and a radically new engine, as compared with those in use today.

And he may not even have to steer the vehicle, -- it will be propelled down a highway merely by keeping it fixed on an electric beam. 11

Others suggest changes just as dramatic. 12

## Mass Transit Forms

However, not all are enamored with the auto. With continued growth in the cities, there appears to be another approach to the carrier operations. 13

Los Angeles is planning two express lines for buses only on a stretch of the San Bernardino Free-way. Mansfield, Ohio, is testing a dial-a-bus service developed by Ford Motor Company; Dallas is trying out electronic devices that control cars seeking to enter expressways; and New York City is expanding its new network of computer-controlled traffic lights. Paris is digging up the Place Vendome and other historic sites for underground parking, and Florence is the latest Italian city to create a pedestrian island, a sizable section of downtown closed to the private car. 14

In Los Angeles, autos are multiplying faster than people. The country's present 3.5 million cars will double in the next decade. 15

The impact of the growing number of cars should be felt most keenly in America's cities. "By the end of this decade, metropolitan traffic volumes are projected to increase roughly 40% in Pittsburgh, 50% in Boston, and 90% in Detroit. 16 Planners predict that traffic will be

moving slower and slower in the future despite regional expenditures already totalling \$3 billion for new high-ways and arterial roads, and \$1.1 billion for a rail rapid-transit system. 17

Freeways were once thought to be the answer but downtown freeway construction is being blocked in city after city.

Some now regard freeways as an enticement to people to drive downtown where ample parking, in many cases, does not exist. "Between 250 and 300 square feet of space is needed for every car that commuters park in the city. Most highway interchanges built to accommodate auto commuters take up at least forty acres of land and sometimes twice as much." 18

Chicago's Streets Commissioner James McDonough indicates:

The greatestanswer to the traffic problem is rapid transit and subways. 19

Other transportation people also believe that coping with today's traffic requires that some demand be shifted from the private auto to public transit.

The American Transit Association figures that "one rail track can move 70,000 persons an hour. One exclusive bus lane can carry 40,000 persons an hour. But only 4,500 motorists can travel on one lane of expressway in an hour using the typical load of one and one-half persons per car." 20

One of the more popular mass transit systems being used in cities around the country is the bus. Milwaukee is a leader in this form of transportation. Milwaukee's buslane now cuts "a 56-minute ride from suburbs to downtown to 33 minutes." Two out of five riders who now ride the bus formerly drove their cars into Milwaukee. They park in fringe lots in shopping centers, then board the bus for quick and efficient transportation. Milwaukee is now planning to build a two-lane highway next to an existing freeway, exclusively for buses. The buslanes are expected to be congestion-free, "carrying 47,000 riders each weekday by 1990, or 25% of the rush-hour travelers who commute along that busy corridor." This

would mean a reduction of about 6,000 in the number of cars that might otherwise be drawn into and parked in Milwaukee. 22

Buses have an important advantage over the fixed rail systems. Not only can they go over 60 miles per hour on private lanes, they can pick up and drop off commuters in downtown sectors of business districts.

Some believe that a better solution is one which combines the convenience of the auto with the high-capacity rate of the rapid transit. One such system is called dual-mode. It would operate in two modes, either manually driven on conventional highways or under automatic control over special guideways.

On the guideway, all driver functions would be performed automatically, with the vehicle picking up its guidance and speed instantaneously from a third rail. The driver could read or relax until his exit. A single guideway 7 or 8 feet wide could carry the same number of cars a 5 lanes of high-way, 10,000 vehicles (or 15,000 people) per hour. 23

The dual-mode system is still filled with many unknowns. General Motors had shied away from developing the dual-mode because the possible damages in case of accident would be so great special legislation limiting liability would be needed. "The fail-safe mechanism to assure safety at high speeds bumper to bumper have yet to be developed and proven." 24

Other cities in the United States are using different types of mass transit systems:

New subways and urban rail lines are being built or definitely planned for construction in San Francisco, Washington, D.C., Atlanta, Baltimore, Los Angeles, Miami, Minneapolis-St. Paul, Pitts-burgh, and Seattle.

Extension of existing rail transit systems are being built or planned-New York, Chicago, Boston, Cleveland, and Philadelphia.

Subway construction is being considered in Buffalo, St. Louis, Dallas, Houston, and Detroit.

Many cities are developing bus routes--lanes which are restricted solely to buses built along-side

existing freeways. These lanes connect large parking facilities in the suburbs and downtown areas. Los Angeles is considering a twelve-mile busway eastward from the city core, making use of a rail right-of-way.

Washington has exclusive bus lanes under construction on Shirley Highway south of the city. When these are completed, the riding time would be cut in half. The city will also "Install a computer to control lights at major intersections throughout the city, and buses will have sensing units that can change the lights to green as they approach an intersection, thus moving mass transit faster."

Pittsburgh is working on two such busways, using a railroad right-of-way and an old trolley car route.

Detroit is acquiring right-of-way for a proposed lane in an express-way median. 27

Howard Sibbley of the Washington Bureau of the World
Herald suggests the viability of the buslane system.

Others have suggested the following alternatives:

- Car pools sponsored by corporation located on stress corridors.<sup>29</sup>
  - 2. Strategic air travel. 30
  - Tube trains. 31
  - 4. Personal rapid transit. 32
  - Automated highways.<sup>33</sup>
- Movable sidewalks, downtown segregation of cars,
   and minibus support service.<sup>34</sup>
- 7. People movers -- smaller than streetcar vehicles usually programmed to conform to the traffic flow which carry people around small cities, large airports and the like. The major problem lies in programming the special demands of individual travelers into the general traffic flow, a necessity if people are to be persuaded to abandon their private cars. 35
- 8. Dial-a-ride -- this is a computerized minibus system, allowing door-to-door service anywhere in town. 36
- Tracked air-cushioned vehicles -- this vehicle
   rides just above the track on a cushion of compressed air.

The vehicle could travel at 150 miles-per-hour initially, and will probably have a feasible speed of 300 miles-per-hour. It might be ideal for awkward areas like southern California, where commuting distances range from 50 to 200 miles -- too long for efficient rail service and too short for big airplanes. 37

Morris Forgash, in his article, "Transportation:

Year 2000," forecasts that locomotives will be nuclearpowered, and the cars and tracks will be wider. Cars will
be cut out, switched (sideways) and loaded and unloaded
automatically by radio-controlled devices. People-pods
will be whished from convenient locations in major cities
and deposited by helicopter on special flat cars to ride
in stream-lined comfort.

We will have advanced highways equipped with radio beams so that drivers can put their vehicles on automatic pilot. Trucks will be banned from city streets.

Monorails in the air, or a "people pipeline" running under ground will relieve the congestion which is choking our cities. 38

# Transportation and Politics

Robert Ayres, an urban transport researcher and Ford Foundation offical in his "Urban Transportation of the Future" suggests that political considerations play an important part of the transit system. He indicates that mass transit is less viable in our economic configuration because it is presently difficult to incorporate into the market system. The auto companies have little incentive to participate, but are still involved in some of the research. Further, the future of urban transportation depends upon which political coalition wins. To make the 'issue understandable, he suggests two "teams":

Now let us look at the characteristics of the urban transportation market: There is an exchange transaction. There is (some) competition. There are buyers and sellers, whom you might call "players", because the market is very much like a "game" in the mathematical sense of the word.

I would like to suggest the oversimplified but not totally inappropriate notion that the players in the market may be viewed as belonging to two teams.

I will call one team red and the other blue.

The red team consists of the automobile industry, the petroleum industry, the state highway departments, the Federal Highway Administration, the construction industry, the cement industry, and most suburbanites -- at least, most suburban politicians.

The blue team is hard to identify for certain.

The U.S. Department of Housing and Urban Development probably belongs on the blue team, along with the Office of Economic Opportunity. Secretary of Transportation Alan Boyd has been talking like a blue. New York Mayor John Lindsay is also a blue as is Ralph Nader. Poor people are definitely blue.

Members of the red team believe generally that low density development of the suburban type is a good thing. They do not like crowds (at least on foot). They believe everybody should have wheels, and mass transit should be left to buses, nothing else. A city's major needs, the red team holds, are more freeways and better parking facilities. High density

development should be discouraged and where it exists should be allowed to die on the vine. No tax money should "subsidize" high density urban centers. High density is bad for people.

The blue team maintains that cities exist, and can't be legislated out of existence. Most of our people live in urban areas, and urban areas without "downtowns" are economic, architectural, and cultural absurdities. Moreover, city living is a well tried and ancient form of himan association. People need a variety of communities and they like diversity of surroundings, access to specialized services, and the amenities that only a city can provide. As far as transportation is concerned, the blue team believes that cities need some freeways to be sure, but also alternatives to freeways and automobiles -- alternatives like minibuses, automated electric taxis, "demandactivated" computer-scheduled jitneys, moving sidewalks, and so on.

The blue team argues, moreover, that in the past cities have paid taxes for highways which benefited

the suburbs and drained the cities of their high income citizens. Suburbanites have run away from their social responsibilities, and now the city needs help.

A forecast of future urban transportation technology is equivalent to a forecast of the outcome of
the game being played between reds and blues. On
the basis of the existing relative economic and pelitical power of these two teams, everybedy in his right
mind would clearly pick red as the winner, although
he perhaps wouldn't want to predict any absolute
victory. 39

From these authors' viewpoint, it would appear that given trend line assumptions, there will be a continuance of auto-dominated systems.

In terms of mass transit, the Public Transportation Assistance Act was passed by Congress in Nevember, 1970, but the funds were not large. 40 The Bill in 1973 to divert funds from highway to mass transit systems failed in the legislative branch of the Federal Government.

# Transportation and Urban Form

How does transportation translate into man-land relationships?

The city according to Cooley emerges because of transportation. 41 The urban structure is a manifestation of at least two major intersects of transportation.

The intersect enhances the city's power in what Cooley called the "power of transfer." That is, the power to transfer goods from one dissimilar transportation mode to another. Latent functions engendered from pwer of transfer include economic power as well as political authority. In terms of Cooley's analysis, the transportation intersect location emerged as a production location.

Razzle, Semple, and others have sought to study the physiographic and topographic aspects to location which cause trasportation breaks, but the intersect theory appears to reamain viable.

Notwithstanding the various methodologies of transportation -- land relationships (land use, trip-purpose regression Markov models), various data has been encountered in this area. 42 Iszard has helped to describe location theory and space preference. 43Huff has suggested that age is the most dominant element in consumer space preference followed by personality, sex, education, mental synthesizing abilities, occupation, and income. Spatial factors play a significant role in determining individual movement patterns like space preference of the individual, his location in the overall space economy, and the relative ease with which he may move from place to place within the system.

However, the important variables in terms of transportation focus around "axial growth." This growth forms
around the major modes of transportation. From analysis
of historical city patterns, various transportation systems
can encourage various forms. Early foot and animal power
encouraged compaction. Later mass transit systems continued impaction, but also encouraged sectoral growth along
the rail corridor. The auto along with the Interstate
and arterial highway has encouraged amoeba-like spread.

Given the continuous use and dominance of the automobile, quasi-colloidal dispersion or sprawl is envisioned.

## Transportation Projections

Given zero-order drift, it would appear quite likely that individual autos (of some form) with its compliment- ary road system shall be the dominant mode of transportation.

In reference to this area, it would appear that the following reasons will keep the system viable. The list does not follow a rank order of importance. Auto travel should prevail because:

- 1. The political coalition can ensure its survival.
- The auto is the safest buffer from crime and racial tension.
- Those most likely to suffer from congestion and pollution near their residence generally do not form a strong coalition.
- 4. Class differences and distances are most likely to extend with the auto as suburbanites can escape and mass transit has the potential to bring large low income groups out to suburban retail outlets and residences.

However, trend line projections would also include some mass transit. But, the overwhelming amount of tax dollars will be spent dealing with the auto and auto-related modes. The other may serve as a support and aesthetic function. The mass transit may be triggered by the city's concern to duplicate, in a small way, what is occurring on a slightly larger level in the megapolitan belt.

Given the drift, it would appear to these authors,
that engineers, planners, and others who normatively and
in an over-technological sense envision future mass transit systems, also overlook facets like the potential military expenditures and war-related activities of the future.
They also appear to miss the importance of this political
coalition and where this coalition is likely to reside.

# A Mass Transit Society

If the following conditions should occur, one might expect a mass transit system. They include: 1) a "green" revolution" of life style, 2) a viable environmental and urban coalition, 3) a move toward post-Keynsian social-

market political and economic system, 4) heavy expenditures in research and development for fast, quick, clean and safer systems, 5) diminishment of crime, 6) racial and class harmony, 7) master land planning and fiscal collectivism.

Accepting the above list, it would appear that the high speed urban mass transit system would emerge. Even then, we still predict some individual autos.

The shape of the city, in terms of population distribution, would tend toward concentration although some horizontal growth would accur along sectoral lines.

### An Interstate System

This projection assumes that certain current problems within the system accelerate and that the highway
system is a direct manifestation of these problems (which
may not be the case). We are indicating here though that
highway development in itself does not necessarily cause
problems nor prevent them, but is one subset of a larger
whole. As an example, fewer interstates would come about

if certain inner-city groups are restricted from movement about the city. However, if the high way system is seen as a viable approach to transport people from the city to the suburbs then more arterials and interstates may be envisioned than even in trend line projections.

Under these, every major city would have an interstate. The auto market would greatly expand. Parking, conjestion, and pollution (dependent upon what kinds of anti-smog devices develop) would present certain problems to the innermetropolitan region, but not necessarily with the same degree to the suburban ring.

City government and services will continue to duplicate and decentralize. The city should have greater spread.

The downtown area would increasingly diminish, but "dollar sovereignty" and the "unseen hand" will have worked its will.

#### Footnotes

- 1 "The Planetary Perspective of John McHale," The Futurist, December, 1970.
  - <sup>2</sup>See manuscript for article.
- <sup>3</sup>Future Directions for Research in Urban Transportation, Report of panel discussion including Sumner Meyers and others. Completed and published under the above title by the Organization for Economic Co-operation and Development (OECD), Publication Office, 2 rue Andre-Pascal, 75 Paris (16<sup>e</sup>), France, 1969.
  - 4 "New Cheaper Products" Newsweek, May 28, 1970.
- <sup>5</sup>Carl H. Madden, "Business and the Future," <u>The Futur-ist</u>, December, 1969.
- <sup>6</sup>Robert Lindsey, "Passengers will Ride to 70's on Hopes and Problems," New York Times, January 11, 1970.
- Jarold A. Keffer, "The Automobile's Success: A Lesson for its Critics," The Futurist, February, 1973.
- Nebraska Department of Roads -- Biennial Report

  1971-72. J.J. Exon, Governor; Thomas D. Doyle, Director -State Engineer.
- 9"TwoThirds of Interstate System Stands Ready for Summer Drivers," National Geographic Society News Bulletin 13:179.
  - 10 ASF Report, 2:2, November, 1968.
- 11 "The Next 65 Years," Financial World, 128:25, October 25, 1967.
- 12 "Wondrous World of 1990: Outlook for Young People,"
  U.S. News & World Report, January 30, 1967, pp. 62-66.

- Walter Cronkite, "Twenty-first Century," Popular Science, 190:98-101, April, 1967.
- 13 "Congress and Urban Mass Transportation," <u>Congress-tonal Digest</u>, 48:289, December, 1969.
- "Untangling Big-City Traffic: The Big Push for Mass Fransit," U.S. News & World Report, 67:48, May 25, 1970.
- 14 "Staving off Auto Paralysis," <u>Business Week</u>, 2113: 54, February 28, 1970.
- <sup>15</sup>Allan T. Demaree, "Cars and Cities on a Collision Course," Fortune, 81:126, February, 1970.
  - 16 Ibid., p. 125.
  - 17 Ibid.
  - 18 Ibid., p. 124.
  - 19 "Auto Paralysis," op. cit.
- 20 "Congress and Urban Mass Transportation, "op. cit., p. 302.
  - "Untangling Big-City Traffic," op. cit.
  - 21 Demaree, op. cit.
  - 22 Ibid., p. 125.
  - 23 Ibid., p. 126.
  - 24 Ibid.
  - 25 "Untangling Big-City Traffic," op. cit.

- "They're Finally Doing Something about Traffic," Changing Times, 24:31, October, 1970.
  - 27 "Untangling Big-City Traffic," op. cit., p. 49.
- "How to Pay for Mass Transit," Business Week, 2095: 72, October 25, 1969.
- 28 Howard Silber, "The People Movers: Busway Experiment Looking Good," World Herald Washington Bureau, Omaha World-Herald, May 30, 1973.
  - 29 "Labor Letter," WSJ, May1, 1973.

News release from Allied Van Lines, Broadview, Illinois, February 28, 1973: "Computerized Study of the Nation's Migration Patterns -- 1972."

"The Energy Crisis: Time for Action," Time, May 7, 1973, p. 41ff.

Keeper, op. cit.

30 Robert Dyment, "Transportation: Wheels, Wings and Hulls for the Year 2000," Science Digest, November, 1968.

Harvey L. Capp, Public Relations, Pan American World Airlines.

- 31 "Piping People," Christian Science Monitor, August 20, 1969.
- 32 David Thompson, "Small Rail Transit 'Ideal for Omaha',"
  Omaha World-Herald, July 4, 1971.

David Thompson, "Mass Transit Study Suggest 4 Routes," Omaha World-Herald, May 23, 1971.

J. Jamison Moore, "What Business Should do NOW to Anticipate the Future," The Futurist, February, 1969. J. Edward Anderson, "PRT: Urban Transportation of the Future?" The Futurist, February, 1973.

"Rapid Transit Instead of Freeway," Omaha World-Herald, April 21, 1971.

"Aust Urges Freeway Studies," Omaha World-Herald, August 13, 1972.

"Rapid Transit Halts Freeway," Omaha World-Herald, November 29, 1971.

David Thompson, "Hearing IS May 8 on Transport Plan," Omaha World-Herald, April 18, 1973.

"Figure on Relocation Actually is \$15,000" Omaha World-Herald, April 18, 1973

"Official: Relocation no Longer Problem," Omaha World-Herald, April 17, 1973.

33 "Transportation for the Bos-Wash Corridor," The Futurist, February, 1972.

Associated Press, Pittsburgh, "Freeway Permit Seen in Future," Christian Science Monitor, March 13, 1968.

34 "The Year 2000 ... Rush Hour," Science Journal, London, (Reprinted in Atlas, May, 1968).

William Marlin, "Future Shock Absorber," Christian Science Monitor, October 30, 1972.

35 "Can \$30 Billion Buy a Better Ride?" Newsweek, 77: 48. January 18, 1971.

36 Ibid.

37 Ibid.

38 Morris Forgash, "Transportation-Year 2000," Modern Transportation: Selected Readings, 1967, p. 415.

"\$30 Billion," loc. cit.

39Robert U. Ayres, "Urban Transpertation of the Future." The Futurist, August, 1968, p. 33.

40 Ibid.

41 Charles H. Cooley, Theory of Transportation (Baltimere, Publications of the American Economic Association, 1894).

42Berry and Horton, "All Consumers Areas."

D. J. Reynolds, Economics, Town Planning and Traffic, Institute of Economic Affairs, 1966.

Wilfred Owen with the assistance of the Insi Bradfield, The Accessible City (The Breekings Institution, 1972).

43op. cit.

David Beeder, "Planner: Sprawl Stumbling Block to Transit Plan," Omaha World-Herald, April 21, 1972.

"Getting There in 1995." Omaha World-Herald, April 18, 197

David England, "Few New Wrinkles," Christian Science Monitor, January 7, 1970.

Chapter Eight

A Summation

#### A Summation

It is tempting, especially after digesting a wide range of forecasts, to draw conclusions about the future. We yield briefly to this urge and relate our thoughts. They are not what we necessarily would like to occur, but rather what we feel probably will happen. We claim not to be prophets, but rather synthesizers utilizing the thoughts of others. These forecasts are molded by significant recurring forces within the system and we recognize how the power of unique events can affect the future.

So, given these assumption, what do we see ahead?

The country should be larger, more centralized, urbanized, and complicated. Specialty subcultures will emerge,
and the family will be more democratic. Religion will not
vanish, but will survive.

Our population should increase moderately and our energy consumption should also increase if it has continued environmental support. We do not envision a pleasant ecological environment, and we see energy shortages, but not enough to cripple a nation.

Politically, our present system should remain, though it may become a bit more authoritarian. We see the continued dominance by the business classes and a few sectors of the labor movement. Corporation capitalism will probably prevail although a few secialist institutions may emerge.

Categorically, the economic forecast appears bright given the country's military dominance. Distribution of the country's wealth will probably still gravitate over-whelmingly to the top.

Although the individual private home will be the ideal, there will be a continued growth of multi-housing in the form of apartments, condominiums, trailer parks and other forms.

The individual auto, although it may change in structure and composition, will remain the most viable mode of transportation.

That is the way we see it, using the perception of others whose wisdom we have attempted to chronicle. Whether it is a "good life" is subject to interpretation.

FINIS